Approved by	Утвержден
Resolution of the sole founder	Решением единственного учредителя
No. [•] dated20	№ [•] ot20_ г.
CHARTER	YCTAB
International Foundation Public Interest Foundation	Международный Фонд "Фонд общественных интересов"

TABLE OF CONTENTS

СОДЕРЖАНИЕ

	Page		Стр.
1.	DEFINED TERMS	1.	ОПРЕДЕЛЕНИЯ3
2.	GENERAL PROVISIONS8	2.	ОБЩИЕ ПОЛОЖЕНИЯ8
3. OF	OBJECTS AND SCOPE OF ACTIVITIES THE FOUNDATION9	3. ФО	ЦЕЛИ И ПРЕДМЕТ ДЕЯТЕЛЬНОСТИ НДА9
4.	ASSETS OF THE FOUNDATION 11	4.	ИМУЩЕСТВО ФОНДА11
5.	MANAGEMENT BODIES 13	5.	ОРГАНЫ ФОНДА13
5.	FORMATION OF THE BOARD 13	6.	ФОРМИРОВАНИЕ СОВЕТА ФОНДА13
7. DE0	COMPETENCE OF THE BOARD. CISION-MAKING PROCEDURE 29	7. ПО	КОМПЕТЕНЦИЯ СОВЕТА ФОНДА. РЯДОК ПРИНЯТИЯ РЕШЕНИЙ29
8.	THE EXECUTIVE DIRECTOR 35	8.	ИСПОЛНИТЕЛЬНЫЙ ДИРЕКТОР 35
9.	THE RIGHTS OF THE FOUNDER 38	9.	ПРАВА УЧРЕДИТЕЛЯ38
10.	CONFLICT OF INTERESTS 39	10.	КОНФЛИКТ ИНТЕРЕСОВ39
	MAKING AMENDMENTS TO THE ARTER41		ПОРЯДОК ВНЕСЕНИЯ ИЗМЕНЕНИЙ СТАВ41
	LIQUIDATION OF THE UNDATION41		ЛИКВИДАЦИЯ ФОНДА41 ПРОЧИЕ ПОЛОЖЕНИЯ42
13.	MISCELLANEOUS 42		ИЛОЖЕНИЕ 1. ТРЕБОВАНИЯ44
SCI	HEDULE 1. REQUIREMENTS 44		ИЛОЖЕНИЕ 2. АНКЕТА47
SCI	HEDULE 2. QUESTIONNAIRE 47		ИЛОЖЕНИЕ 3. СОГЛАСИЕ50
SCI	HEDULE 3. CONSENT 50	ПРІ	ИЛОЖЕНИЕ 4. КОМПЕТЕНЦИЯ
ВО	HEDULE 4. COMPETENCE OF THE ARD. DECISION-MAKING	CO	ВЕТА ФОНДА И ПОРЯДОК ПРИНЯТИЯ ЦЕНИЙ51
	OCEDURE51	ПРІ	ИЛОЖЕНИЕ 5. ОСОБЫЙ ПОРЯДОК ИНЯТИЯ РЕШЕНИЙ63
SCE	HEDULE 5. SPECIFIC REQUIREMENTS DECISION-MAKING63		
SCI	HEDULE 6. FORM OF THE POWER OF TORNEY76		ИЛОЖЕНИЕ 6. ФОРМА ВЕРЕННОСТИ76

1.	DEFINED TERMS	1.	ОПРЕДЕЛЕНИЯ
1.1.	Questionnaire has the meaning defined in paragraph 6.15 of this Charter	1.1.	Анкета имеет значение, указанное в п. 6.15 Устава.
1.2.	Interim General Director has the meaning as set forth in the Yandex LLC's Charter.	1.2.	Временный Генеральный директор имеет значение, указанное в Уставе ООО Яндекс.
1.3.	General Director has the meaning as set forth in the Yandex LLC's Charter.	1.3.	Генеральный директор имеет значение, указанное в Уставе ООО Яндекс.
1.4.	Yandex Group means the Founder and all persons under the Founder's Control.	1.4.	Группа Яндекс означает Учредителя и лиц, находящихся под Контролем Учредителя.
1.5.	Applicable Law has the meaning defined in paragraph 2.2 of this Charter.	1.5.	Действующее законодательство имеет значение, указанное в п. 2.2 Устава.
1.6.	Director A has the meaning defined in paragraph 6.2.1 of this Charter.	1.6.	Директор А имеет значение, указанное в п. 6.2.1 Устава.
1.7.	Director B has the meaning defined in paragraph 6.2.2 of this Charter.	1.7.	Директор В имеет значение, указанное в п. 6.2.2 Устава.
1.8.	Director C has the meaning defined in paragraph 6.2.3 of this Charter.	1.8.	Директор С имеет значение, указанное в п. 6.2.3 Устава.
1.9.	Additional Grounds shall mean a list of additional grounds for termination of the employment agreement with General Director set out in Appendix 5 to Yandex LLC's Charter.	1.9.	Дополнительные основания означает список дополнительных оснований для прекращения трудового договора с Генеральным Директором, в соответствии с Приложением 5 к Уставу ООО Яндекс.
1.10.	Law means [Federal Law No. 290-FZ "On international companies and international foundations".]	1.10.	Закон означает [Федеральный закон № 290-ФЗ "О международных компаниях и международных фондах".]
1.11.	Executive Director means the sole executive body of the Foundation.	1.11.	Исполнительный директор означает единоличный исполнительный орган Фонда.
1.12.	ITMO means Federal State Autonomous Educational Institution of Higher Education "National Research University ITMO", registration No. 1187746579690.	1.12.	ИТМО означает федеральное государственное автономное образовательное учреждение высшего образования «Национальный исследовательский университет ИТМО», ОГРН 1187746579690.

- 1.13. **Public Policy Committee** means Public Policy Committee of the board of directors of the Founder, formed in accordance with the Founder's Articles and governed by a charter adopted in accordance with the Founder's Articles.
- 1.14. **Nominating Committee** shall mean the nominating committee of the board of directors of the Founder, formed in accordance with the Founder's Articles and governed by a charter adopted in accordance with the Founder's Articles.
- 1.15. **Control** means, in relation to any entity: 1.15. (a) direct or indirect authority to determine and procure determination of the management decisions in relation to such entity (due to holding any securities, or partnership rights, or other types of holding, under a contract or otherwise); (b) entitlement to directly or indirectly determine voting by over 50% votes at any general meeting (or an equivalent body) of such entity; or (c) the right to appoint over 50% of members of a board (or any closest equivalent) of such entity. "Controlling", "Controlled" and "under Control" shall be construed accordingly.

1.16. **Ownership Cap** means the restriction on holding shares in the Founder in accordance with the definition of the Ownership Cap under the Founder's Articles.

- 1.13. Комитет по общественным интересам означает комитет по общественным интересам при совете директоров Учредителя, сформированный в соответствии с Уставом Учредителя и осуществляющий свою деятельность на основании положения, утвержденного в соответствии с Уставом Учредителя.
- 1.14. Комитет по назначениям означает комитет по назначениям директоров при совете директоров Учредителя, сформированный в соответствии с Уставом Учредителя и осуществляющий свою деятельность на основании положения, утвержденного в соответствии с Уставом Учредителя.
 - Контроль означает по отношению к любому юридическому лицу: (а) наличие прямых или опосредованных полномочий определять или обеспечивать определение управленческих решений такого юридического лица (на основании владения ценными бумагами или правами партнера, или иными правами собственности, на основании договора или на любом другом основании); (b) возможность прямо или опосредованно распоряжаться более чем 50% (пятьюдесятью процентами) голосов на любых общих собраниях (или в эквивалентных органах) такого юридического лица; или (с) возможность назначать более 50% (пятидесяти процентов) состава совета директоров (или ближайшего эквивалента такому органу управления) такого юридического лица, а слова "Контролирующий", "Контролируемый" и "находящийся под Контролем" должны толковаться в соответствии с настоящим определением.
- 1.16. **Лимит владения** означает ограничение на владение акциями Учредителя в соответствии с определением *Ownership Cap* по Уставу Учредителя.

- 1.17. **MSU** means Federal State Budget Educational Institution of Higher Education M.V. Lomonosov Moscow State University, registration No. 1037700258694.
 - государственное бюджетное образовательное учреждение высшего образования "Московский государственный университет имени М.В. Ломоносова", ОГРН 1037700258694.

МГУ означает федеральное

1.17.

- 1.18. **MIPT** means Federal State
 Autonomous Educational Institution of
 Higher Education "Moscow Institute of
 Physics and Technology (National
 Research Institute)", registration No.
 1027739386135.
- 1.18. **МФТИ** означает федеральное государственное автономное образовательное учреждение высшего образования "Московский физико-технический институт (национальный исследовательский университет)", ОГРН 1027739386135.
- 1.19. **HSE** means Federal State Autonomous Educational Institution of Higher Education "National Research University "Higher School of Economics", registration No. 1027739630401.
- 1.19. **НИУ ВШЭ** означает федеральное государственное автономное образовательное учреждение высшего образования "Национальный исследовательский университет "Высшая школа экономики", ОГРН 1027739630401.
- 1.20. **Notary** means any one of the notaries on the List of Notaries.
- 1.20. **Нотариус** означает любого из нотариусов, включенных в Список нотариусов.
- 1.21. **Binding Nomination** shall mean a nomination by the Foundation as the holder of the Priority Share of a candidate to serve as Designated Director, which shall be made in accordance with the Founder's Articles and internal documents of the Founder.
- 1.21. Обязывающая номинация означает предложение Фондом как владельцем Специальной акции кандидатуры на должность Специального директора в порядке, предусмотренном Уставом и внутренними документами Учредителя.
- 1.22. **Yandex LLC** means Yandex Limited Liability Company, registration No. 1027700229193.
- 1.22. **ООО Яндекс** означает Общество с ограниченной ответственностью Яндекс, ОГРН 1027700229193.
- 1.23. **Special Corporate Situation** has the meaning as defined by Yandex LLC's Charter.
- 1.23. Особая корпоративная ситуация имеет значение, указанное в Уставе ООО Яндекс.
- 1.24. **Special Situation** has the meaning as defined by Yandex LLC's Charter.
- 1.24. **Особая ситуация** имеет значение, указанное в Уставе ООО Яндекс.
- 1.25. **RSPP** means the All-Russia Public Organization "Russian Union of Industrialists and Entrepreneurs", Registration No. 1037700126540.
- 1.25. РСПП означает Общероссийскую общественную организацию "Российский союз промышленников и предпринимателей", ОГРН 1037700126540,

1.26. **Secretary** means the person that acts as 1.26. Секретарь означает лицо, the secretary of the Board meeting. осуществляющее функции секретаря заседания Совета Фонда. 1.27. 1.27. Foundation Secretary means the Секретарь Фонда означает corporate secretary of the Foundation. корпоративный секретарь Фонда. 1.28. 1.28. **AV Family Trust** means certain trust Семейный траст АВ означает траст, established by Arkady Yurievich учрежденный Аркадием Юрьевичем Volozh existing under the laws of Воложем, существующий по Cayman Islands. законодательству Каймановых островов. 1.29. **CEO** means the Chief Executive 1.29. СЕО означает главный Officer of the Founder within the исполнительный директор (Chief meaning of the Founder's Articles. Executive Officer) Учредителя, как этот термин используется в Уставе Учредителя. 1.30. 1.30. Skolkovo means Non-Governmental Сколково означает Educational Institution of Additional негосударственное образовательное Professional Education Moscow School учреждение дополнительного of Management "SKOLKOVO", профессионального образования Московская Школа Управления registration No. 1075000012405. "СКОЛКОВО", ОГРН 1075000012405. **Board of Directors of the Founder** 1.31. 1.31. Совет директоров Учредителя означает совет директоров means board of directors of the Founder acting on the basis of the Founder's Учредителя, действующий на Articles and internal documents of the основании Устава Учредителя и Founder. внутренних документов Учредителя. **Board** means the collegial management 1.32. 1.32. Совет Фонда означает body of the Foundation. коллегиальный орган управления Фонда. 1.33. **Consent** has the meaning defined in 1.33. Согласие имеет значение, указанное paragraph 6.9 of this Charter в п. 6.9 Устава. SPbGU means Federal State Budgetary 1.34. 1.34. СПбГУ означает федеральное Educational Institution of Higher государственное бюджетное Education "Saint-Petersburg State образовательное учреждение University", registration No. высшего образования "Санкт-1037800006089 Петербургский государственный университет", ОГРН 1037800006089 1.35. **Priority Share** means Priority Share in 1.35. Специальная акция означает the share capital of the Founder, which специальную акцию (Priority Share) в grants the rights and powers stipulated акционерном капитале Учредителя, in the Founder's Articles. объем прав и полномочий по которой регулируется Уставом Учредителя. 1.36. **Special Interest** means a participatory 1.36. Специальная доля означает долю в interest in the charter capital of Yandex уставном капитале ООО Яндекс LLC with a nominal value of 166.05 номинальной стоимостью 166,05

(one hundred sixty six and 05/100) rubles, which accounts for 0.001% (1/1000 percent) of Yandex LLC charter capital, contributed to the Foundation by the Founder.		рублей (сто шесть десят шесть рублей 5 копеек), что составляет 0.001% (ноль целых одна тысячная процента) уставного капитала ООО Яндекс, переданную в имущество Фонда Учредителем.
Designated Director shall mean a member of the board of directors of the Founder defined as Designated Director in the Founder Articles.	1.37.	Специальный директор означает члена совета директоров Учредителя, который обозначен как специальный директор (<i>Designated Director</i>) в соответствии с Уставом Учредителя.
List of Notaries means a list of notaries that may be approved and amended by the Board from time to time in accordance with this Charter.	1.39.	Список нотариусов означает список нотариусов, который время от времени утверждается и изменяется Советом Фонда в соответствии с Уставом.
List of Candidates has the meaning as defined in Yandex LLC's Charter.	1.38.	Список кандидатов имеет значение, указанное в Уставе ООО Яндекс.
Requirements means the requirements specified in <u>Schedule 1</u> to the Charter.	1.39.	Требования означает требования, установленные в <u>Приложении 1</u> к Уставу.
University means any one of the organizations listed in paragraph 6.2.2 of the Charter.	1.40.	Университет означает любую из организацией, указанных в п. 6.2.2 Устава.
Charter means version of the Foundation's charter, as in effect from time to time.	1.41.	Устав означает редакцию устава Фонда, действующей на соответствующую дату.
Yandex LLC's Charter means the charter of Yandex LLC, as in effect from time to time.	1.42.	Устав ООО Яндекс означает устав ООО Яндекс в редакции, действующей на соответствующую дату.
Founder's Articles means the articles of association of the Founder, as in effect from time to time.	1.43.	Устав Учредителя означает устав Учредителя в редакции, действующей на соответствующую дату.
Founder has the meaning defined in paragraph 2.1 of the Charter.	1.44.	Учредитель имеет значение, указанное в п. 2.1 Устава.
Foundation has the meaning defined in paragraph 2.1 of the Charter.	1.45.	Фонд имеет значение, указанное в п. 2.1 Устава
School № 57 Development Fund means Fund for the aid and promotion of the development of the educational and scientific activities of State	1.46.	Фонд поддержки Пятьдесят седьмой школы означает Фонд поддержки и содействия развитию образовательной научной

1.37.

1.38.

1.40.

1.41.

1.42.

1.43.

1.44.

1.45.

1.46.

1.47.

1.48.

and scientific activities of State

Budgetary General Education Institution

образовательной, научной, просветительской деятельности

государственного бюджетного

of Moscow "School № 57", registration No. 1187700018747.

- общеобразовательного учреждения города Москвы "Пятьдесят седьмая школа", ОГРН 1187700018747.
- 1.49. **Expert** means any of PwC, EY,
 Deloitte or KPMG (or their subsidiaries or branches operating in the territory of the Russian Federation) or such other auditor company, as may be approved by the Board from time to time.
- 1.47. Эксперт означает одну из аудиторских компаний PwC, EY, Deloitte или KPMG (или их дочерние общества или подразделения, действующие на территории России) или иную аудиторскую компанию, которая может быть одобрена Советом Фонда.
- 1.50. **Expert Institution** means any one of the organizations listed in paragraph 6.2.3 of the Charter.
- 1.48. Экспертная организация означает любую из организаций, указанных в п. 6.2.3 Устава.

2. GENERAL PROVISIONS

2. ОБЩИЕ ПОЛОЖЕНИЯ

- 2.1. International Foundation "Public Policy Foundation" (the "**Foundation**") is a unitary non-commercial organization without membership established by Yandex N.V., registered in accordance with the laws of the Netherlands in the Trade Register under the registration number 27265167 (the "**Founder**").
- 2.1. Международный фонд "Фонд Общественных Интересов" ("Фонд") является не имеющей членства унитарной некоммерческой организацией, учреждённой Компанией с ограниченной ответственностью Яндекс Н.В. (Yandex N.V.), зарегистрированной по праву Нидерландов в Торговом реестре за регистрационным номером 27265167 ("Учредитель").
- 2.2. The Foundation operates in accordance with the Law, other laws and regulatory legal acts of the Russian Federation ("Applicable Law") and this Charter.
- 2.2. Фонд осуществляет свою деятельность в соответствии с Законом, иными законами и нормативными правовыми актами Российской Федерации ("Действующее законодательство") и настоящим Уставом.
- 2.3. The full name of the Foundation in Russian: Международный Фонд "Фонд общественных интересов".
- 2.3. Полное наименование Фонда на русском языке: Международный Фонд "Фонд общественных интересов".
- 2.4. The abbreviated name of the Foundation in Russian: MΦ "ΦΟΗ".
- 2.4. Сокращенное наименование Фонда на русском языке: МФ "ФОИ".
- 2.5. The full name of the Foundation in English: International Foundation Public Interest Foundation.
- 2.5. Полное наименование Фонда на английском языке: International Foundation Public Interest Foundation
- 2.6. The abbreviated name of the Foundation in English: IF PIF.
- 2.6. Сокращенное наименование Фонда на английском языке: IF PIF.
- 2.7. The Foundation is considered formed and acquires the rights and obligations
- 2.7. Фонд считается созданным и приобретает права и обязанности

of a legal entity from the moment of its state registration. The foundation is established for an indefinite period.

- 2.8. The Foundation owns separate assets and is liable for its obligations with these assets, has its separate balance sheet, has the right to acquire and exercise property and non-property rights in its own name, assume obligations, act as plaintiff and defendant in court, open settlement, foreign currency and other accounts in banking and credit institutions.
- 2.9. The Foundation is liable for its obligations with all of its assets capable of being collected for debts in accordance with Applicable Law. The Founder is not liable for the obligations of the Foundation, nor is the Foundation liable for the obligations of its Founder. The Foundation is not liable for the obligations of the state, nor is the state liable for the obligations of the Foundation.
- 2.10. Registered address of the Foundation: Russian Federation, [Kaliningrad].

3. OBJECTS AND SCOPE OF ACTIVITIES OF THE FOUNDATION

- 3.1. The Foundation is established to perform managerial functions, and its statutory object is to preserve the continuity and promote the success of the Yandex Group business.
- 3.2. In connection with its objects the Foundation performs the following activities:
- 3.2.1. ownership of the assets transferred to the Foundation by the Founder and exercise of the rights and fulfillment of the obligations associated with the ownership of such assets in accordance with this Charter, including:

юридического лица с момента его государственной регистрации. Фонд создается без ограничения срока деятельности.

- 2.8. Фонд имеет в собственности обособленное имущество и отвечает по своим обязательствам этим имуществом, имеет самостоятельный баланс, вправе от своего имени приобретать и осуществлять имущественные и неимущественные права, нести обязанности, быть истцом и ответчиком в суде, открывать расчетные, валютные и иные счета в банковских и кредитных организациях.
- 2.9. Фонд отвечает по своим обязательствам всем принадлежащим ему имуществом, на которое может быть обращено взыскание в соответствии с Действующим законодательством. Учредитель не отвечает по обязательствам Фонда, а Фонд не отвечает по обязательствам своего Учредителя. Фонд не несет ответственности по обязательствам государства, а государство не несет ответственности по обязательствам Фонда
- 2.10. Местонахождение Фонда: Российская Федерация, [г. Калининград].

3. ЦЕЛИ И ПРЕДМЕТ ДЕЯТЕЛЬНОСТИ ФОНДА

- 3.1. Фонд учрежден в управленческих целях, и его уставной целью является сохранение преемственности и успешное развитие Группы Яндекс.
- 3.2. Для достижения своих уставных целей Фонд осуществляет следующие виды деятельности:
- 3.2.1. владение имуществом, переданным Фонду в собственность Учредителем, и осуществление прав и обязанностей, связанных с владением таким имуществом, в соответствии с настоящим Уставом, включая:

- (a) ownership of the Priority Share; and exercising the rights and fulfilling obligations of the Priority Share holder in accordance with the laws applicable to the Priority Share on the terms set forth in the Founder's Articles; and
- (b) ownership of the Special Interest and exercising the rights and fulfilling obligations of a participant in Yandex LLC, as provided for under the Applicable Law and Yandex LLC's Charter;
- 3.2.2. exercise of other corporate rights and powers in connection with the ownership of the Priority Share and the Special Interest, in the scope set out under the applicable laws and the Founder's Articles and Yandex LLC's Charter accordingly,

in each case, only to achieve the statutory objects of the Foundation and only in the manner stipulated by the Founder and set out in this Charter.

- 3.3. The Foundation has the right to pursue the activities that directly contribute to the attainment of the objects specified in paragraph 3.1 of this Charter.
- 3.4. The Foundation may not engage in income-generating activities using the assets referred to in paragraph 3.2.1 of this Charter. In accordance with the Law, the exercise of the rights and the receipt of income under the Priority Share and the Special Interest, as well as participation in commercial entities is not recognized as income-generating activities (commercial activities).
- 3.5. In relation to the Priority Share, the Foundation is considered a qualified investor upon its contribution to the Foundation.

- (а) владение Специальной акцией и осуществление прав и обязанностей владельца Специальной акции в соответствии с применимым к Специальной акции правом на условиях, установленных Уставом Учредителя; и
- (б) владение Специальной долей и осуществление прав и обязанностей участника ООО Яндекс, предусмотренных Действующим законодательством и уставом ООО Яндекс;
- 3.2.2. осуществление иных корпоративных прав и полномочий в связи с владением указанными Специальной акцией и Специальной долей в объеме, предусмотренном применимым законодательством, Уставом Учредителя и Уставом ООО Яндекс, соответственно,

в каждом случае строго в целях соблюдения уставных целей Фонда и в порядке, предусмотренном Учредителем и установленном в настоящем Уставе.

- 3.3. Фонд вправе осуществлять виды деятельности, непосредственно связанные с целями, указанными в п. 3.1 Устава.
- 3.4. Фонд не вправе осуществлять приносящую доход деятельность с использованием имущества, указанного в п. 3.2.1 Устава. В соответствии с Законом, осуществление прав и получение дохода по Специальной акции и Специальной доле, а равно участие в хозяйственных обществах не признается приносящей доход деятельностью (предпринимательской деятельностью).
- 3.5. Фонд является квалифицированным инвестором в отношении Специальной акции с момента ее передачи в имущество Фонда.

4. ASSETS OF THE FOUNDATION

- 4.1. The Foundation has the right to own or have other property rights in any assets to which civil rights may apply, including monetary funds in rubles and in foreign currencies, securities and other assets.
- 4.2. The assets of the Foundation shall be formed from the following sources:
- 4.2.1. one-time and regular contributions from the Founder:
- 4.2.2. contributions to finance the Foundation and its statutory activities;
- 4.2.3. income payable on the assets owned by the Foundation, including dividends (income, interest) payable on the shares, debentures and other securities, and on deposits;
- 4.2.4. other earnings not prohibited by Applicable Law.
- 4.3. Other than the members of Yandex Group, no third parties are allowed to make contributions to finance the Foundation.
- 4.4. Income generated through the activities of the Foundation shall not be distributed to the Founder and shall be used only for financing the statutory activities of the Foundation, except for the distribution of the assets of the Foundation to the Founder upon liquidation of the Foundation in the manner provided for in paragraph 12.3 of the Charter.
- 4.5. On the basis of part 2 of Article [12⁷] of the Law, the Foundation does not have the right to dispose of its assets, including those specified in paragraph 3.2.1 of the Charter, as follows:
- 4.5.1. alienate the Priority Share and Special Interest, including by way of exit from Yandex LLC or requesting Yandex LLC to acquire its Special Interest;

4. ИМУЩЕСТВО ФОНДА

- 4.1. Фонд вправе иметь в собственности или на ином вещном праве любые объекты гражданских прав, в том числе денежные средства в рублях и иностранной валюте, ценные бумаги и иное имущество.
- 4.2. Источниками формирования имущества Фонда являются:
- 4.2.1. единовременные и регулярные поступления (взносы) от Учредителя;
- 4.2.2. поступления на содержание Фонда и ведение им уставной деятельности;
- 4.2.3. доходы, получаемые от имущества, находящегося в собственности Фонда, в том числе дивиденды (доходы, проценты), получаемые по акциям, облигациям, другим ценным бумагам и вкладам;
- 4.2.4. другие поступления, не запрещенные Действующим законодательством.
- 4.3. За исключением лиц, входящих в Группу Яндекс, пополнение имущества Фонда третьими лицами не допускается.
- 4.4. Доходы от осуществления деятельности Фонда не распределяются Учредителю Фонда и должны использоваться только для ведения уставной деятельности Фонда, за исключением распределения имущества Фонда Учредителю при ликвидации в порядке, предусмотренном п. 12.3 Устава.
- 4.5. На основании ч. 2 ст. [12⁷] Закона, Фонд не вправе распоряжаться своим имуществом, в том числе указанным в п. 3.2.1 Устава, следующими способами:
- 4.5.1. отчуждать Специальную акцию и Специальную долю, в том числе путем реализации права на выход из ООО Яндекс или направления

требования о приобретении ООО Яндекс Специальной доли;

- 4.5.2. transfer the Priority Share and the Special Interest under a trust management with the right of the manager to exercise the rights attached thereto;
- 4.5.2. передавать Специальную акцию и Специальную долю в доверительное управление с правом осуществления прав по указанным Специальной акции и Специальной доле;
- 4.5.3. charge any assets as a security under any obligations of the Foundation or a third party;
- 4.5.3. предоставлять какое-либо имущество в качестве обеспечения по каким-либо обязательствам Фонда или третьего лица;
- 4.5.4. provide rights of gratuitous use over any 4.5.4. assets of the Foundation;
- 4.5.4. предоставлять права безвозмездного пользования имуществом Фонда;
- 4.5.5. alienate any exclusive rights over intellectual property or grant any exclusive license in relation to intellectual property.
- 4.5.5. отчуждать исключительные права на результаты интеллектуальной деятельности или предоставлять исключительную лицензию на использование результатов интеллектуальной деятельности.

Information about the restrictions imposed by this paragraph, as well as provisions of paragraph 12.3 shall be recorded with the Unified Federal Register of Legally Significant Information about the Activities of the Legal Entities and Entrepreneurs and Other Subjects of Economic Activities. Any transaction made by the Foundation in violation of such restriction shall be null and void (part 1 of Art. 174¹ of the Civil Code).

Сведения о наличии ограничений, предусмотренных настоящим пунктом, а также условия, предусмотренные п. 12.3, подлежат внесению в Единый федеральный реестр юридически значимых сведений о фактах деятельности юридических лиц, индивидуальных предпринимателей и иных субъектов экономической деятельности. Сделка, совершенная Фондом в нарушение указанного ограничения, является ничтожной (часть 1 статьи 1741 Гражданского кодекса).

- 4.6. Certain types of transactions of the Foundation require prior approval by the Founder and (or) Board in accordance with paragraphs 8.5 and 9.2 or other provisions of this Charter.
- 4.6. Отдельные виды сделок Фонда требуют получения предварительного согласия Учредителя и (или) Совета Фонда на их совершение в соответствии с пп. 8.5 и 9.2 или иными положениями настоящего Устава.
- 4.7. The Founder has the right to make, at any time, one-time or regular contributions to the assets of the Foundation on the basis of a unilateral transaction that does not require the approval of the Foundation or of any other person. As of the date of the approval of this version of the Charter, the Founder issues to the Foundation an
- 4.7. Учредитель вправе в любое время передавать единовременные и регулярные поступления (взносы) в имущество Фонда на основании односторонней сделки, не требующей получения согласия Фонда или какого-либо иного лица. В дату утверждения настоящей редакции Устава Учредитель предоставляет

irrevocable guaranty to cover Foundation's ongoing expenses.

Фонду безотзывную гарантию в обеспечение обязательств Фонда по финансированию его текущих расходов.

5. MANAGEMENT BODIES

5. ОРГАНЫ ФОНДА

- 5.1. The Foundation shall have the following management bodies:
- 5.1. Органами Фонда являются:

5.1.1. the Board; and

5.1.1. Совет Фонда; и

- 5.1.2. the Executive Director.
- 5.1.2. Исполнительный директор.
- 5.2. The management bodies of the Foundation shall act within their scope of competence in accordance with this Charter.
- 5.2. Органы Фонда действуют в пределах своей компетенции в соответствии с настоящим Уставом.
- 5.3. The main function of the Board is to ensure that the Foundation achieves the objects for which it was established.
- 5.3. Основной функцией Совета Фонда является обеспечение соблюдения Фондом целей, для которых он создан.
- 5.4. The Executive Director is responsible for day-to-day management of the Foundation's activities and acts on behalf of the Foundation without a power of attorney.
- 5.4. Исполнительный Директор осуществляет текущее руководство деятельностью Фонда и действует от имени Фонда без доверенности.

6. FORMATION OF THE BOARD

6. ФОРМИРОВАНИЕ СОВЕТА ФОНДА

- 6.1. The Board shall consist of eleven (11) members, all of which shall be natural persons.
- 6.1. Совет Фонда состоит из 11 (одиннадцати) членов физических лиц.
- 6.2. The Board shall be formed following the procedure set out below:
- 6.2. Совет Фонда формируется в соответствии с порядком, установленным ниже:
- 6.2.1. three members of the Board that include the individual that is serving as the CEO from time to time (or if such office is then vacant, another executive director of the Founder), together with two members of the management bodies of the Founder or any entity within Yandex Group as he/she may appoint for this purpose. Hereinafter the Board members appointed in accordance with this paragraph 6.2.1 are together referred to as "Directors A", and "Director A" means any one of them;
- 6.2.1. три члена Совета Фонда, которые включают лицо, занимающее в соответствующий момент времени должность СЕО (а в случае если такое лицо не назначено, другого исполнительного директора Учредителя) и двух назначенных им лиц, являющихся членами органов управления Учредителя или какоголибо юридического лица Группы Яндекс. Далее члены Совета Фонда, назначенные в соответствии с данным п. 6.2.1 совместно именуются

"Директора А", и "Директор А" означает любого из них;

6.2.2.

- 6.2.2. five members of the Board that include:
 - (a) one individual nominated by MSU;
 - (b) one individual nominated by MIPT;
 - (c) one individual nominated by HSE:
 - (d) one individual nominated by ITMO; and
 - (e) one individual nominated by SPbGU.

hereinafter the Board members appointed in accordance with this paragraph 6.2.2 are together referred to as "**Directors B**", and "**Director B**" means any one of them. During the entire term of their duties, Directors B shall satisfy the Requirements;

- 6.2.3. three members of the Board that include:
 - (a) one individual nominated by RSPP; and
 - (b) one individual nominated by Skolkovo:
 - (c) one individual nominated by School № 57 Development Fund;

hereinafter the Board members appointed in accordance with this paragraph 6.2.3 are together referred to as "**Directors C**", and "**Director C**" means any one of them. Directors C shall satisfy the Requirements (save for the instances when failure to satisfy the Requirements is permissible in accordance with second subsection of paragraph 6.11);

- пять членов Совета Фонда, которые являются:
 - (а) лицом, предложенным МГУ;
 - (б) лицом, предложенным МФТИ;
 - (в) лицом, предложенным НИУ ВШЭ;
 - (г) лицом, предложенным ИТМО; и
 - (д) лицом, предложенным СПбГУ;

далее члены Совета Фонда, назначенные в соответствии с данным п. 6.2.2 совместно именуются "Директора В", и "Директор В" означает любого из них. Директора В в течение всего срока своих полномочий должны соответствовать Требованиям;

- 6.2.3. три члена Совета Фонда, которые являются:
 - (а) лицом, предложенным РСПП; и
 - (б) лицом, предложенным Сколково;
 - (в) лицом, предложенным Фондом поддержки Пятьдесят седьмой школы;

далее члены Совета Фонда, назначенные в соответствии с данным п. 6.2.3 совместно именуются "Директора С", и "Директор С" означает любого из них. Директора С в течение всего срока своих полномочий должны соответствовать Требованиям (за исключением случая, когда несоответствие Требованиям допускается в соответствии с положениями второго абзаца п. 6.11);

- 6.2.4. for the purposes this Charter Directors A, Directors B and Directors C form separate classes of Board members.
- 6.3. The initial Board shall be formed by the Founder by the resolution approving this version of the Charter. The decision on the approval of this version of the Charter specifies: (i) the class of Board members under paragraph 6.2 above, to which each so appointed Board member belongs; (ii) the University or Expert Institution by which such Board member is deemed to be nominated, if applicable; and (iii) contact details and other information for sending notices to the members of the initial Board, as well as to the Founder, Universities and Expert Institutions. Except as otherwise provided in paragraph 6.12 of this Charter, after the formation of the initial Board of the Foundation, the Board shall be formed on the basis of the principles established in paragraph 6.2 of the Charter.

- 6.4. The list of the Universities and Expert Institutions as set forth in this Charter may not be changed, except by amendment of the Charter with a Founder's prior approval under paragraph 9.2.1 of this Charter.
- 6.5. The term of office of the members of the Board is unlimited. The termination of the powers of one member of the Board shall not entail the termination of powers of all the other members of the Board. An incomplete Board shall retain its powers, but is not authorized to decide on the following matters:
- 6.5.1. amendment of the Charter:

- 6.2.4. далее для целей настоящего Устава Директора А, Директора В и Директора С составляют отдельные классы членов Совета Фонда.
- 6.3. Первоначальный состав Совета Фонда формируется Учредителем при утверждении настоящей редакции Устава. Решение об утверждении настоящей редакции Устава содержит указание о том, (і) к какому классу членов Совета Фонда, из указанных в п. 6.2 выше, относится каждый назначенный член Совета Фонда, (іі) по предложению какого Университета или Экспертной организации он считается назначенным (если применимо), и (ііі) контактные данные и реквизиты для направления уведомлений членам Совета Фонда, составляющим первоначальный состав Совета Фонда, а также Учредителю, Университетам и Экспертным организациям. За исключением случая, предусмотренного п. 6.12 настоящего Устава, после формирования первоначального состава Совета Фонда состав Совета Фонда формируется исходя из принципов, установленных п. 6.2 Устава.
- 6.4. Состав Университетов и Экспертных организаций, установленный в настоящем Уставе, не подлежит изменению, кроме как в порядке изменения Устава с согласия Учредителя в соответствии с п. 9.2.1 Устава.
- 6.5. Срок полномочий членов Совета Фонда не ограничен. Прекращение полномочий одного члена Совета Фонда не влечет прекращения полномочий всех остальных членов Совета Фонда. В случае если Совет Фонда не сформирован полностью, он сохраняет свою компетенцию, но не вправе принимать какие-либо из следующих решений:
- 6.5.1. внесение изменений и дополнений в Устав;

- 6.5.2. liquidation of the Foundation;
- 6.5.3. to the extent the number of incumbent Directors A is less than two, any decisions that require at least two votes of Directors A.
- 6.6. The powers of a Board member shall be terminated on the following grounds:
- 6.6.1. voluntary resignation of the Board member, in which case such Board member notifies the Foundation with a copy to all incumbent members of the Board and the relevant University or the Expert Institution, which nominated (or is deemed to have nominated) such Board member, if applicable. The powers of the Board member shall be terminated from the moment of such notification to the Foundation, unless another date is indicated in the notification;
- 6.6.2. death of the Board member, recognition of him/her as missing, dead, legally incompetent or insolvent in the manner prescribed by any law applicable to such person in any jurisdiction. The powers of such Board member shall be terminated on the day of his/her death or on the date of the court decision that recognizes him/her as missing, dead, legally incompetent, insolvent, as the case may be, comes into effect;
- 6.6.3. if Director B or Director C ceases to satisfy any of the Requirements then the powers of Director B or Director C shall be terminated:

- 6.5.2. ликвидация Фонда;
- 6.5.3. в случае, если среди действующих членов Совета Фонда менее двух Директоров А, какие-либо вопросы, для принятия решений по которым требуется как минимум два голоса Директоров А.
- 6.6. Полномочия члена Совета Фонда прекращаются по следующим основаниям:
- 6.6.1. в случае добровольного сложения с себя полномочий членом Совета Фонда, о чем соответствующий член Совета Фонда направляет уведомление Фонду с копией всем действующим членам Совета Фонда и соответствующему Университету или Экспертной организации, в зависимости от того, кто предложил кандидатуру (кто считается предложившим кандидатуру) такого члена Совета Фонла. Полномочия соответствующего члена Совета Фонда прекращаются с момента получения Фондом уведомления о сложении полномочий, если в самом уведомлении не указана иная дата;
- 6.6.2. смерти члена Совета Фонда, признания его в порядке, установленным каким-либо применимым правом в любой юрисдикции, безвестно отсутствующим, умершим или недееспособным, а также несостоятельным (банкротом). Полномочия соответствующего члена Совета Фонда прекращаются в день его смерти или в день вступления в законную силу решения суда о признании его безвестно отсутствующим, умершим, недееспособным или несостоятельным, соответственно;
- 6.6.3. в случае если какой-либо Директор В или Директор С перестает соответствовать какому-либо из Требований, полномочия соответствующего Директора В или Директора С прекращаются:

- (a) on the date when the relevant Director B or Director C notifies the Foundation (with a copy to the Founder) of non-compliance with the Requirements, including by way of submitting the Questionnaire as envisaged by paragraph 6.15; or
- (b) if Director B or Director C does not give a notice of noncompliance with the Requirements, then:
 - (i) upon expiry of twenty (20) calendar days from the Founder's, Director A's or Director B's notice to the Foundation (with a copy to the relevant Director B or Director C) of the noncompliance by such Director B or Director C with the Requirements, which specifies with reasonable level of detail the basis for such noncompliance; or
 - (ii) if Director B or Director C ceases to satisfy the Requirements solely because of a commercial conflict of interest as defined in item 5 of Schedule 1, which is caused by the Yandex Group's actions (including as a result of acquisition of any new business by Yandex Group), upon six (6) months from the Founder's notice to the Foundation (with a copy to the relevant Director B or Director C) of the noncompliance of the relevant Director B or Director C with the Requirements, which specifies with reasonable level of detail the basis for such noncompliance; and

- (а) в дату уведомления Директором В или Директором С Фонда (с копией Учредителю) о своем несоответствии Требованиям, в том числе путем направления Анкеты в порядке, предусмотренном п. 6.15; или
- (б) в отсутствие уведомления Директора В или Директора С о своем несоответствии Требованиям:
 - (i) по истечении 20 (двадцати) календарных дней с даты направления Учредителем, Директором А или Директором В уведомления Фонду (с копией такому Директору В или Директору С) о несоответствии Директора В или Директора С Требованиям с указанием в разумно необходимом объеме причин такого несоответствия; или
 - (ii) в случае если Директор В или Директор С перестает соответствовать Требованиям исключительно в связи с возникновением у него коммерческого конфликта интересов, как этот термин определен в п. 5 Приложения 1, в результате действий Группы Яндекс (включая в результате приобретения Группой Яндекс нового бизнеса), по истечении 6 (шести) месяцев с даты направления Учредителем уведомления Фонду (с копией такому Директору В или Директору С) о несоответствии такого Директора В или Директора С Требованиям с указанием в разумно необходимом объеме

причин такого несоответствия; и

unless in each case specified in paragraphs 6.6.3(b)(i) and 6.6.3(b)(ii), prior to the expiry of the relevant term set forth in paragraph 6.6.3(b)(i) or 6.6.3(b)(ii), the relevant Director B or Director C, who was identified as non-compliant by the Founder's, Director A's or Director B's notice, has not sent to the Foundation (with a copy to the Founder) a written objection with reasonable level of detail as to why such Director B or Director C considers himself/herself compliant with the Requirements. Upon receipt of such objection by Director B or Director C, the Founder within 3 (three) months from the date of receipt of a written objection by the relevant Director B or Director C may at its discretion appoint an Expert for the purposes of determining whether the Board member is compliant with the Requirements, and the powers of the relevant Director B or Director C shall be terminated on the date of the receipt by the Foundation (with a copy to such Director B or Director C) of the written opinion of the Expert confirming that such Director B or Director C does not satisfy the Requirements. If the Founder has not provided the Foundation (with a copy of such Director B or Director C) the written opinion of the Expert confirming that such Director B or Director C does not satisfy the Requirements within 3 (three) months from the date of receipt of the written objection of the relevant Director B or Director C, termination of powers of the respective Director B or Director C is permissible only subject to a new notice by the Founder submitted in the manner as

в каждом указанном в пп. 6.6.3(б)(і) и 6.6.3(б)(іі) случае при условии, что до истечения соответствующего срока, установленного в пп. 6.6.3(б)(i) или 6.6.3(б)(іі) выше, соответствующий Директор В или Директор С, в отношении которого Учредитель, Директор А или Директор В направил уведомление о несоответствии Требованиям, не направил в Фонд (с копией Учредителю) письменное возражение с разумно необходимым обоснованием того, почему он полагает себя соответствующим Требованиям. В случае получения такого возражения от Директора В или Директора С Учредитель по своему усмотрению вправе в течение 3 (трех) месяцев с даты получения письменного возражения соответствующего Директора В или Директора С привлечь Эксперта для определения соответствия члена Совета Фонда Требованиям, и полномочия соответствующего Директора В или Директора С подлежат прекращению в дату получения Фондом письменного заключения Эксперта (с копией такому Директору В или Директору С), содержащего подтверждение того, что указанный Директор В или Директор С не соответствует Требованиям. Если Учредитель не предоставил Фонду (с копией такому Директору В или Директору С) письменное заключение Эксперта, содержащее подтверждение того, что указанный Директор В или Директор С не соответствует Требованиям в течение 3 (трех) месяцев с даты получения письменного возражения соответствующего

provided for by paragraphs 6.6.3(b)(i) or 6.6.3(b)(ii)) above.

Директора В или Директора С, прекращение полномочий соответствующего Директора В или Директора С возможно при условии направления Учредителем нового уведомления о его несоответствии Требованиям в порядке, предусмотренном пп. 6.6.3(б)(і) или 6.6.3(б)(іі) выше.

For avoidance of doubt, the Founder, Director A or Director B is entitled to appoint an Expert prior to notifying the Foundation (with a copy to the relevant Director B or Director C) of the non-compliance by Director B or Director C with the Requirements as envisaged by paragraphs 6.6.3(b)(i) or 6.6.3(b)(ii), in which case the powers of Director B or Director C shall terminate upon expiry of twenty (20) calendar days or upon expiry six (6) months from the relevant notice, as applicable in each case.

In the circumstances specified in paragraphs 6.6.3(a) and 6.6.3(b), the Foundation shall be obliged to immediately notify the University or the Expert Institution that has nominated (is deemed to have nominated) the relevant Director B or Director C of the receipt of the Founder's notice regarding their non-compliance with the Requirements.

- 6.6.4. if the person serving as the CEO replaces any other Director A. The powers of Director A terminate from the appointment of a new Director A as set forth by paragraph 6.9.1 of the Charter;
- 6.6.5. if the University replaces Director B who was nominated (or is deemed nominated) by such University. The powers of Director B terminate from the

Во избежание сомнений Учредитель, Директор А или Директор В вправе также привлечь Эксперта до направления уведомления Фонду (с копией такому Директору В или Директору С) о несоответствии Директора В или Директора С Требованиям в порядке, предусмотренном пп. 6.6.3(б)(і) или 6.6.3(б)(іі) выше, и в таком случае полномочия Директора В или Директора С прекращаются по истечении 20 (двадцати) календарных дней или по истечении 6 (шести) месяцев с даты направления соответствующего уведомления, в зависимости от обстоятельств.

В случаях, указанных в пп. 6.6.3(а)и 6.6.3(б) выше Фонд обязуется незамедлительно уведомить Университет или Экспертную организацию, которые предложили кандидатуру (считаются предложившими кандидатуру) Директора В или Директора С, в отношении которого получено уведомление о несоответствии Требованиям, о получении такого уведомления.

- 6.6.4. в связи с решением лица, занимающего должность СЕО, о замене Директора А. Полномочия Директора А прекращаются с момента назначения нового Директора А в порядке, предусмотренном п. 6.9.1 Устава;
- 6.6.5. в связи с решением Университета, по предложению которого назначен (считается назначенным) Директор В, о замене такого Директора В.

appointment by the University of a new Director B as set forth by paragraphs 6.9.2 of the Charter;

- 6.6.6. if the Expert Institution replaces
 Director C who was nominated (or is deemed nominated) by such Expert
 Institution. The powers of Director C terminate from the appointment by the Expert Institution of a new Director C as set forth by paragraphs 6.9-6.11 of the Charter;
- 6.6.7. if a decision to terminate his/her appointment is made by all the other incumbent members of the Board unanimously (e.g. in the event a member of the Board does not comply with the requirements of the Charter, fails to fulfill decisions of the Board. fails to fulfill his/her duties as a member of the Board, does not attend three (3) consecutive meetings of the Board). The powers of such Board member are terminated from the date of the unanimous decision of all the other incumbent Board members, unless another date is indicated in their decision.
- 6.7. Other than envisaged by paragraph 6.8 below, from the time of the termination of the powers of a member of the Board and until the time of the appointment of a new member of the Board as his/her replacement in accordance with paragraphs 6.9-6.11, the vote of the terminated member is not taken into account for the purpose of decision-making by the Board.
- 6.8. Notwithstanding the provisions of paragraph 6.7, to the extent permitted by the Applicable Law:
- 6.8.1. in the event the powers of Director A are terminated for any reason, other than envisaged by paragraph 6.6.4, until a new Director A is appointed in accordance with paragraph 6.9.1, the voting right of the terminated Director A shall be temporarily transferred to

Полномочия Директора В прекращаются с момента назначения Университетом нового Директора В порядке, предусмотренном пп. 6.9.2 Устава:

- 6.6.6. в связи с решением Экспертной организации, по предложению которой назначен (считается назначенным) Директор С, о замене такого Директора С. Полномочия Директора С прекращаются с момента назначения Экспертной организацией нового Директора С в порядке, предусмотренном пп. 6.9-6.11 Устава;
- 6.6.7. по единогласному решению всех остальных действующих членов Совета Фонда (в том числе вследствие несоблюдения членом Совета Фонда требований Устава, неисполнения решений Совета Фонла, неисполнения своих обязанностей в качестве члена Совета Фонда, неучастие в 3 (трех) подряд заседаниях Совета Фонда). Полномочия соответствующего члена Совета Фонда прекращаются с даты принятия единогласного решения остальными действующими членами Совета Фонда, если иная дата не указана в таком решении.
- 6.7. За исключением случаев, указанных в п. 6.8 ниже, с момента прекращения полномочий члена Совета Фонда и до назначения нового члена Совета Фонда взамен прекратившего полномочия в соответствии с положениями пп. 6.9-6.11 голос такого прекратившего полномочия члена Совета Фонда при голосовании не учитывается.
- 6.8. Несмотря на положения п. 6.7, в той степени, в которой это допускается Действующим законодательством:
- 6.8.1. в случае прекращения полномочий Директора А по любым основаниям, кроме предусмотренных в п. 6.6.4, до назначения нового Директора А взамен прекратившего полномочия в соответствии с п. 6.9.1, право голоса выбывшего Директора А временно

another Director A, provided that any incumbent Director A may exercise voting rights of not more than one terminated Director A and no longer than within sixty (60) days from the termination of such Director A. The voting right of the terminated Director A shall be transferred to the Director A holding the position of the CEO, or in case of CEO's termination, to the Director A who has been serving on the Board for the longer time period, and if such time period is the same – to the older one;

- in the event the powers of Director B are terminated for any reason, other than envisaged by paragraph 6.6.5, until a new Director B is appointed, in accordance with paragraph 6.9.2, the voting right of the terminated Director B shall be temporarily transferred to another Director B, provided that any incumbent Director B may exercise voting rights of not more than one terminated Director B and no longer than within sixty (60) days from the termination of such Director B. The voting right of the terminated Director B shall be transferred to the incumbent Director B nominated (considered to be nominated) by the University appearing first above on the list in paragraph 6.2.2, who is not exercising the voting rights of another terminated Director B;
- 6.8.3. in the event the powers of Director C are terminated for any reason, other than envisaged by paragraph 6.6.6, until a new Director C is appointed, in accordance with paragraph 6.11, the rights and obligations of the relevant Director C shall be temporarily vested with the most senior officer of the Expert Institution, which nominated (or is deemed to have nominated) Director C whose powers are so terminated. The relevant Expert Institution, and

передается другому Директору А при условии, что каждый действующий Директор А может осуществлять право голоса не более чем одного выбывшего Директора А и в течение не более 60 (шестидесяти) дней с момента выбытия соответствующего Директора А. Право голоса выбывшего Директора А передается Директору А, являющемуся СЕО, а в случае прекращения полномочий СЕО – такому Директору А, который в течение более долгого срока занимает должность в Совете Фонда, и если такой срок является одинаковым – более старшему по возрасту;

- 6.8.2. в случае прекращения полномочий Директора В по любым основаниям, кроме предусмотренных в п. 6.6.5, до назначения нового Директора В взамен прекратившего полномочия в соответствии с п. 6.9.2, право голоса выбывшего Директора В временно передается другому Директору В при условии, что каждый действующий Директор В может осуществлять право голоса не более чем одного выбывшего Директора В и в течение не более 60 (шестидесяти) дней с момента выбытия соответствующего Директора В. Право голоса выбывшего Директора В передается действующему Директору В, кандидатура которого была предложена (считается предложенной) Университетом, первым по очередности из указанных в п. 6.2.2 Университетов, который не осуществляет право голоса другого выбывшего Директора В;
- 6.8.3. в случае прекращения полномочий Директора С по любым основаниям, кроме предусмотренных в п. 6.6.6, до назначения нового Директора С взамен прекратившего полномочия в соответствии с п. 6.11, права и обязанности такого Директора С временно исполняет высшее должностное лицо Экспертной организации, по предложению которой был назначен (считался назначенным) Директор С,

Directors A and Directors B, shall endeavor to fill such position within 30 days of such vacancy arising. Within such term of 30 days, the relevant most senior officer of the Expert Institution may perform the duties of Director C irrespective of whether he/she satisfies the Requirements and whether a consent by the Board has been granted in accordance with paragraph 6.10.2. Upon expiry of such 30-day period, any of Director A or Director B shall be entitled to initiate the termination of the powers of such most senior officer of the Expert Institution serving as Director C on the ground of his/her noncompliance with the Requirements in accordance with the procedure set out in paragraph 6.6.3(b);

прекративший полномочия. Соответствующая Экспертная организация, Директора А и Директора В должны приложить необходимые усилия для согласования кандидата на вакантную позицию Директора С в течение 30 дней с момента прекращения полномочий такого Директора С. В течение указанных 30 дней соответствующее высшее должностное лицо Экспертной организации вправе осуществлять права и обязанности Директора С независимо от того, соответствует ли указанное лицо Требованиям и было ли получено согласие Совета Фонда на его назначение Директором С в порядке, указанном в п. 6.10.2. По истечении такого срока любой из Директоров А и Директоров В вправе инициировать прекращение полномочий такого высшего должностного лица Экспертной организации в качестве Директора С по основанию его несоответствия Требованиям в порядке, предусмотренном п. 6.6.3(б);

- 6.8.4. in the event a Director B is not attending a Board meeting of the Board held in accordance with paragraph 7.20 in person, such Director B may authorize another incumbent Director B to exercise his/her voting rights at the Board meetings under a power of attorney issued in form set out in Schedule 6 for a period of not more than thirty (30) days from the date of issue. Such power of attorney must be certified by a notary.
- 6.8.4. в случае, если какой-либо из Директоров В не принимает личное участие в заседании Совета Фонда, проводимого в соответствии с п. 7.20, такой Директор В вправе уполномочить другого действующего Директора В распоряжаться своим голосом на заседаниях Совета Фонда на основании доверенности, составленной по форме Приложения 6 и выданной на срок не более, чем 30 (тридцать) календарных дней с даты выдачи. Такая доверенность подлежит нотариальному удостоверению.
- 6.8.5. in the event any member of the Board is not attending a meeting of the Board held in accordance with paragraph 7.15 in person, such member of the Board may authorize another incumbent member of the Board of the same class to exercise his/her voting rights at the Board meetings under a power of attorney issued in form set out in Schedule 6 for a period of not more than
- 6.8.5. в случае, если какой-либо из членов Совета Фонда не принимает личное участие в заседании Совета Фонда, проводимого в соответствии с п. 7.15, такой член Совета Фонда вправе уполномочить другого действующего члена Совета Фонда того же класса распоряжаться своим голосом на заседаниях Совета Фонда на основании доверенности,

fifteen (15) calendar days from the date of issue. Such power of attorney must be certified by a notary; and

- 6.9. Upon occurrence of any ground for termination of the powers of a Board member as specified in paragraphs 6.6.1-6.6.3 or for the purposes of replacing a member of the Board as envisaged in paragraphs 6.6.4-6.6:
- 6.9.1. in relation to Director A, the person serving as the CEO replaces Director A, whose powers have been terminated or who is proposed to be replaced by appointment of a new Director A;
- 6.9.2. in relation to Director B, the relevant University, which nominated (or is deemed to have nominated) such terminated Director B, replaces Director B, whose powers have terminated or who is proposed to be replaced; and
- 6.9.3. in relation to Director C: the relevant Expert Institution, which nominated (or is deemed to have nominated) such terminated Director C, shall nominate candidate(s) to replace Director C, whose powers have been terminated or who is proposed to be replaced;

in each case by written notice to the Foundation (with a copy to the incumbent members of the Board) containing information about the appointed Director A or Director B or candidate(s) for the position of Director C, as relevant.

In case of appointment of Director A or Director B in accordance with paragraphs 6.9.1-6.9.2, such appointment becomes effective immediately upon the written notice to the Foundation. In case of appointment of Director B or nomination of candidates to the positions of Director C

составленной по форме Приложения 6 и выданной на срок не более, чем 15 (пятнадцать) календарных дней с даты выдачи. Такая доверенность подлежит нотариальному удостоверению; и

- 6.9. В случае возникновения какого-либо из оснований прекращения полномочий члена Совета Фонда, указанных в пп. 6.6.1 6.6.3 или в целях замены члена Совета Фонда в соответствии с пп. 6.6.4-6.6.6:
- 6.9.1. в отношении Директора А, лицо, занимающее должность СЕО, назначает Директора А взамен прекратившего полномочия или подлежащего замене;
- 6.9.2. в отношении Директора В, соответствующий Университет, по предложению которого назначен (считается назначенным) Директор В, прекративший полномочия, назначает Директора В взамен прекратившего полномочия или подлежащего замене; и
- 6.9.3. в отношении Директора С: соответствующая Экспертная организация, по предложению которой назначен (считается назначенным) Директор С, прекративший полномочия, предлагает кандидата(ов) на должность Директора С взамен прекратившего полномочия или подлежащего замене:

в каждом случае путем письменного уведомления в адрес Фонда (с копией действующим членам Совета Фонда) с информацией о назначенном Директоре А или Директоре В и кандидате(ах) на позицию Директора С, соответственно.

В случае назначения или замены какого-либо Директора А или Директора В в соответствии с пп. 6.9.1-6.9.2 соответствующее назначение вступает в силу в дату письменного уведомления Фонда о таком назначении. В случае назначения Директора В или

in accordance with paragraphs 6.9.2-6.9.3, the relevant written notice shall include documents as reasonably necessary to confirm that (i) each candidate for the position of Director B or Director C meets the Requirements, including a filled out Questionnaire, and (ii) each candidate has agreed to be appointed as a member of the Board, which shall be evidenced by a written consent in form of Schedule 3 (the "Consent"). Any nomination of the Director must be accompanied by nomination document duly executed by an authorized representative of the Founder, the University or the Expert Institution, as applicable.

- предложения кандидатов на должность Директора С в соответствии с пп. 6.9.2-6.9.3, письменное уведомление должно включать документы, разумно необходимые для подтверждения: (i) соответствия каждого кандидата на должность Директора В или Директора С Требованиям, в том числе заполненную Анкету такого кандидата, и (ii) наличия согласия каждого кандидата на назначение членом Совета Фонда, выраженного в письменной форме в соответствии с Приложением 3 ("Согласие"). Любое предложение кандидатов на должность Директоров Фонда должно быть предоставлено в виде уведомления о предложении кандидата, подписанного уполномоченным лицом от имени лица, Учредителя, Университета или Экспертной организации, как применимо.
- 6.10. Not later than two (2) business days from the date of the receipt by the Foundation of the notification containing information about the candidate(s) nominated to the position of Director C in accordance with paragraph 6.9.3, the Executive Director or another member of the Board shall set the date for a meeting of the Board, the agenda of which includes (i) the verification of the compliance of the proposed candidates to the position of Director C with the Requirements and (ii) the granting of consent to their appointment. Such meeting of the Board shall be held within the term set forth in paragraph 7.7. Following the consideration by the Board of the above matters:
- 6.10.1. a candidate shall be deemed to meet the Requirements if at least two Directors A and at least two Directors B vote "in favor" on the matter of compliance of the candidate with the Requirements;
- 6.10.2. the consent to the appointment of the candidate is deemed granted if at least two of Directors A and at least two of

and

- 6.10. Не позднее 2 (двух) рабочих дней с даты получения Фондом уведомления о кандидатах, предложенных на должность Директора С в соответствии с п. 6.9.3, Исполнительный директор или иной член Совета Фонда назначает дату проведения заседания Совета Фонда, в повестку дня которого входят вопросы (і) о проверке соответствия заявленных кандидатов на должность Директора С Требованиям и (іі) предоставлении согласия на их назначение. Такое заседание Совета Фонда должно быть проведено в сроки, предусмотренные п. 7.7. По итогам рассмотрения Советом Фонда указанных вопросов:
- 6.10.1. кандидат признается соответствующим Требованиям, если как минимум два Директора А и как минимум два Директора В проголосовали "за" по вопросу о соответствии кандидата Требованиям; и
- 6.10.2. согласие на назначение кандидата считается предоставленным, если не менее 2 Директоров А и не менее 2

Directors B vote "in favor" on the matter of granting a consent by the Board to the appointment of such candidate to the position of Director C.

The Board shall notify the person who has nominated the relevant candidate(s) to the position of a Board member of the decision made by the Board within one (1) business day from the execution date of the minutes of the meeting in accordance with the notification procedure set out herein.

6.11. From the date the candidate nominated to the position of Director C is granted a consent to the appointment in accordance with paragraph 6.10.2, such candidate is appointed to the position of Director C.

In the event a candidate to the position of Director C is not deemed to meet the Requirements, but a consent to his/her appointment was received in accordance with paragraphs 6.10.2, and such candidate was appointed to the position of Director C as set out in this paragraph 6.11, the Founder is entitled to initiate the termination of the powers of such Director C at any time in accordance with paragraph 6.6.3 and notwithstanding the consent to his appointment provided such person does not meet (continues not to meet) the Requirements after being appointed to the position of Director C.

In the event that in relation to none of the candidates proposed for the position of Director C by an Expert Institution the Board has granted its consent to such candidate's appointment, then the relevant nominating Expert Institution shall have the right to propose other candidates in accordance with the same procedure as described in paragraph 6.9.3.

6.12. If at any time voting power over securities carrying at least 50% plus one of the votes under all outstanding

Директоров В проголосовали "за" по вопросу о предоставлении согласия Совета Фонда на назначение такого кандидата Директором С.

Информация о принятом Советом Фонда решении доводится до лица, предложившего кандидата(ов) в члены Совета Фонда, не позднее чем в течение 1 (одного) рабочего дня с даты составления протокола заседания в порядке, предусмотренном Уставом для уведомлений.

6.11. С даты принятия в отношении кандидата на должность Директора С решения о предоставлении согласия на его назначение в порядке, указанном в п. 6.10.2, данный кандидат назначается Директором С.

Если в отношении какого-либо кандидата на должность Директора С, не было принято решение о его соответствии Требованиям, но было предоставлено согласие на его назначение в порядке, предусмотренном пп. 6.10.2, и такой кандидат был назначен на должность Директора С в порядке, указанном в настоящем п. 6.11, то Учредитель вправе инициировать прекращение полномочий Директора С в соответствии с п. 6.6.3, несмотря на данное ранее согласие на его назначение, если он не соответствует Требованиям (продолжает не соответствовать Требованиям) после его назначения Директором С.

В случае, если в отношении ни одного из кандидатов, предложенных на должность Директора С Экспертной организацией, не было принято решение о предоставлении согласия Совета Фонда на его назначение, соответствующая Экспертная организация вправе предложить других кандидатов в соответствии с процедурой, установленной в п. 6.9.3.

6.12. В случае если в какой-либо момент времени права голоса по ценным бумагам, предоставляющим как

ordinary shares in the Founder is not held cumulatively (directly or indirectly) by one or more individuals holding Russian citizenship and (or) persons that are under Control of such individuals, state corporations, state companies, public-law companies, state and municipal enterprises, establishments and organizations, as well as their Controlled persons and other entities Controlled by the state, federal authorities, authorities of the constituent subjects of the Russian Federation or municipal authorities, then:

минимум 50% всех голосов плюс один голос по всем выпущенным обыкновенным акциям Учредителя, не принадлежат в совокупности (прямо или опосредованно) одному или нескольким физическим лицам, имеющим российское гражданство, и (или) лицам, находящимся под Контролем таких физических лиц, государственным корпорациям, государственным компаниям. публично-правовым компаниям, государственным и муниципальным предприятиям, учреждениям и организациям, а равно их Контролируемым лицам и иным организациям, Контролируемым государством, органами федерального управления, органами субъектов Российской Федерации или органами муниципального управления, то:

- 6.12.1. there shall be two (2) Directors A on the Board. The person that is authorized to appoint Directors A in accordance with paragraph 6.2.1 shall make a decision on the termination of the powers of a particular Director A and shall notify the Foundation thereof. The powers of such Director A shall be terminated from the date of such notification. To the extent the person referred to in paragraph 6.2.1 does not make a decision on termination of the powers of a particular Director A within thirty (30) days from the moment specified in paragraph 6.12, the Board shall be entitled to make such decision. The relevant decision by the Board shall be adopted by the majority of votes of Directors B;
- 6.12.1. в составе Совета Фонда должно быть 2 (два) Директора А. Лицо, которое имеет право назначать Директоров А в соответствии с п. 6.2.1, принимает решение о прекращении полномочий конкретного Директора А, и уведомляет о принятом решении Фонд. Полномочия соответствующего члена Совета Фонда прекращаются в дату соответствующего уведомления. Если лицо, указанное в п. 6.2.1, не принимает решение о прекращении полномочий конкретного Директора А в течение 30 (тридцати) дней с момента, указанного в п. 6.12, то решение о том, полномочия какого Директора А подлежат прекращению, принимает Совет Фонда. Такое решение Совета Фонда принимается большинством голосов Директоров В;
- 6.12.2. HSE appoints one additional Director B to the Board in accordance with the procedure set out in paragraph 6.9.2, so that after such appointment there are six (6) Directors B. All provisions of this Charter attributable to Directors B shall apply to the additional Director B in full; and
- 6.12.2. НИУ ВШЭ назначает дополнительного Директора В в соответствии с процедурой назначения, предусмотренной п. 6.9.2, таким образом, чтобы после его назначения среди членов Совета Фонда было 6 (шесть) Директоров В. Все положения Устава, применимые к Директорам В, применяются к

дополнительному Директору В полном объеме:

6.12.3. The Board will continue to include three 6.12.3. в члены Совета Фонда по-прежнему (3) Directors C.

For the purposes of this paragraph 6.12, the below persons AV Family Trust at all times shall be deemed to be under Control of individuals holding Russian citizenship.

If at any time after the composition of the Board is amended in accordance with this paragraph 6.12, the voting power over securities carrying the voting power over securities carrying the number of votes under all outstanding ordinary shares in the Founder as specified in paragraph 6.12 above will be again held cumulatively (directly or indirectly) by the persons as listed in paragraph 6.12, then: (i) the Board shall be formed in accordance with the principles described in paragraph 6.2 of the Charter from the moment the Founder notifies the Foundation of the relevant circumstances; (ii) on the date of the receipt of such notification by the Foundation, the powers of Director B nominated in accordance with paragraph 6.12.2 shall terminate; and (iii) the person referred to in paragraph 6.2.1 is entitled to appoint third Director A in accordance with the procedure set out in paragraph 6.9.1 of the Charter.

- 6.13. The procedure for the formation of the Board provided for by this Charter cannot be changed, except by unanimous decision of all members of the Board and upon a prior approval of the Founder.
- 6.14. The members of the Board shall not be considered employees of the Foundation. The members of the Board may receive remuneration and compensation of expenses directly related to participation on the Board, provided that the relevant amounts should not exceed the annual amount of

входят 3 (три) Директора С.

Для целей настоящего п. 6.12 указанные ниже лица Семейный траст АВ всегда считается лицом, находящимися под Контролем физических лиц, имеющих российское гражданство.

В случае если после изменения состава Совета Фонда в соответствии с настоящим п. 6.12, в какой-либо момент права голоса по ценным бумагам, предоставляющим право голоса в отношении указанного выше в п. 6.12 количества голосов по всем выпущенным обыкновенным акциям Учредителя, будут снова принадлежать в совокупности (прямо или опосредованно) лицам, перечисленным в п. 6.12, то: (і) Совет Фонда подлежит формированию в соответствии с принципами, указанными в п. 6.2 Устава с момента, когда Учредитель уведомляет Фонд о соответствующих обстоятельствах; (ii) в дату получения такого уведомления Фондом полномочия Директора В, назначенного в соответствии с п. 6.12.2 прекращаются, и (ііі) лицо, указанное в п. 6.2.1, вправе назначить третьего Директора А в соответствии с процедурой назначения, предусмотренной пп. 6.9.1 Устава.

- 6.13. Порядок формирования Совета Фонда, предусмотренный настоящим Уставом, не может быть изменен, кроме как по единогласному решению всех членов Совета Фонда и с предварительного согласия Учредителя.
- 6.14. Члены Совета Фонда не являются работниками Фонда. Членам Совета Фонда может выплачиваться вознаграждение и компенсация расходов, непосредственно связанных с участием в работе Совета Фонда, в размере, не превышающем годовой размер таких

such payments as set out in the financial plan (budget) approved by the Board with the Founder's consent.

- 6.15. Directors B and Directors C on an annual basis are obliged to submit to the Foundation with a copy to the Founder a filled out questionnaire concerning their compliance with the Requirements in form of Schedule 2 hereto (the "Questionnaire"), as well as notify the Foundation of any changes in the information set out in the Questionnaire by submitting to the Foundation with a copy to the Founder of a new filled out Questionnaire not later than within three (3) business days after such changes occurred.
- 6.16. Board members are obliged to notify the Foundation (with a copy to all other Board members) of their contact details and information for the purposes of notices, requests and communications sent in accordance with this Charter in each case of changes to the previously notified details. At the request of any Board member, the Executive Director or Foundation Secretary are obliged to provide such Board member with the contact details and information of all other Board members, as well as Universities and Expert Institutions.
- 6.17. Directors B and Directors C are obliged to notify the Foundation of contact details and information of the Universities and Expert Intuitions that have appointed (are deemed to have appointed) for notice purposes in each case of changes to the previously notified details.
- 6.18. Prior to receipt by the Foundation of any notice from a Board member of a change in contact details or information of such Board member, the relevant University or Expert Institution, all notices addressed as notified previously

- выплат по финансовому плану (бюджету) Фонда, одобренному Советом Фонда и согласованному с Учредителем.
- 6.15. Директора В и Директора С обязаны раз в год предоставлять в Фонд с копией Учредителю заполненную анкету касательно соответствия Требованиям по форме Приложения 2 ("Анкета"), а также уведомлять Фонд в случае изменения каких-либо сведений, указанных в Анкете, путем направления Фонду с копией Учредителю вновь заполненной Анкеты не позднее чем в течение 3 (трех) рабочих дней после возникновения таких изменений.
 - Члены Совета Фонда обязаны уведомлять Фонд (с копией другим членам Совета Фонда) о своих контактных данных и реквизитах для целей направления уведомлений, требований и иных сообщений в соответствии с Уставом в каждом случае их изменения. По запросу любого члена Совета Фонда Исполнительный директор или Секретарь Фонда обязаны предоставить такому члену Совета Фонда информацию о контактных данных и реквизитах всех других членов Совета Фонда, а также Университетов и Экспертных организаций.
- 6.17. Директора В и Директора С обязаны информировать Фонд о контактных данных и реквизитах для направления уведомлений в адрес Университетов и Экспертных организаций, которые их назначили (считаются назначившими), в каждом случае их изменения по сравнению с данными, ранее сообщенными Фонду.
- 6.18. До даты получения Фондом уведомления члена Совета Фонда об изменении каких-либо контактных данных и реквизитов такого члена Совета Фонда, соответствующего Университета или соответствующей Экспертной организации, все

shall be deemed to be properly addressed.

уведомления, направленные по предыдущим реквизитам, считаются направленными надлежащим образом.

7. COMPETENCE OF THE BOARD. DECISION-MAKING PROCEDURE

7. КОМПЕТЕНЦИЯ СОВЕТА ФОНДА. ПОРЯДОК ПРИНЯТИЯ РЕШЕНИЙ

- 7.1. The Board is entitled to decide on the matters referred to its competence under the Applicable Law and this Charter. The decisions of the Board shall be passed in accordance with the procedures set out in this section 7 and Schedule 4 hereto.
- 7.1. Совет Фонда принимает решения по вопросам, отнесенным к его компетенции Действующим законодательством и Уставом. Порядок принятия решений Советом Фонда установлен в настоящем разделе 7 и в Приложении 4 к Уставу.
- 7.2. Issues referred to the competence of the Board cannot be referred to any other management bodies of the Foundation.
- 7.2. Вопросы, отнесенные к компетенции Совета Фонда, не могут быть переданы для решения другим органам Фонда.
- 7.3. The members of the Board shall elect a chair from among themselves. The chair of the Board shall be elected and replaced by a simple majority vote of the Board members. The chair is elected for an indefinite term.
- 7.3. Из состава членов Совета Фонда избирается председатель Совета Фонда. Решение об избрании или смене председателя Совета Фонда принимается простым большинством голосов членов Совета Фонда. Председатель Совета Фонда избирается на неопределенный срок.
- 7.4. Meetings of the Board are held at least once a year. An ordinary meeting of the Board is held no earlier than three (3) and no later than six (6) months after the end of each calendar (financial) year.
- 7.4. Заседания Совета Фонда проводятся не реже 1 (одного) раза в год. Очередное заседание Совета Фонда проводится в срок не ранее 3 (трех) и не позднее 6 (шести) месяцев после окончания календарного (финансового) года.
- 7.5. An extraordinary meeting of the Board is convened by the Executive Director on his own initiative or at the request of any member of the Board. If a Board meeting is called to consider any matters listed in items 10 and 11(a)-(h) of Schedule 4, such Board meeting may be convened by any Director B.
- 7.5. Внеочередное заседание Совета Фонда созывается Исполнительным директором по собственной инициативе или по требованию любого члена Совета Фонда. Заседания Совета Фонда по вопросам, перечисленным в п. 10 и 11(а)-(з) Приложения 4, могут созываться любым Директором В.
- 7.6. A request to convene an extraordinary meeting of the Board shall be signed by the persons requesting its convocation and sent to the Executive Director together with the list of such persons and the proposed agenda.
- 7.6. Требование о созыве внеочередного заседания Совета Фонда направляется за подписью его инициаторов Исполнительному директору с указанием имен инициаторов, а также предлагаемой повестки дня созываемого

внеочередного заседания Совета Фонда.

- 7.7. The Executive Director shall, within two (2) business days from the date of the receipt of the request to convene an extraordinary meeting of the Board, set a date for its holding. An extraordinary meeting of the Board shall be convened and held within four (4) business days from the day the Executive Director receives a request to convene such meeting. In the event no decision to convene a Board meeting is made within the term set forth above, as well as in the event specified in paragraph 7.5, the Board meeting may be convened by the Board member who has requested convocation of such Board meeting.
- 7.8. The Executive Director, or another person who is convening the Board meeting in accordance with paragraph 7.7, shall notify the members of the Board of the convocation of the meeting and of the agenda and shall e-mail to them all materials and information required for preparing for the meeting at least two (2) business days before the date of the meeting. For the avoidance of doubt, the Executive Director shall notify the members of the Board of the convocation of the meeting and of the agenda immediately following the making of decision on the date for holding a meeting as per paragraph 7.7.
- 7.9. Members of the Board shall have the right to propose additional issues for the agenda within one (1) business day after they have been notified of the convocation of a meeting of the Board. If additional issues are included in the agenda, the members of the Board shall be notified of the fact at least one (1) business day before the date of the meeting.

- 7.7. Исполнительный директор обязан не позднее 2 (двух) рабочих дней со дня получения требования о созыве внеочередного заседания Совета Фонда назначить дату его проведения. Внеочередное заседание Совета Фонда должно быть созвано и проведено в течение 4 (четырех) рабочих дней со дня получения Исполнительным директором требования о созыве такого внеочередного заседания Совета Фонда. Если в течение указанного срока решение о проведении заседания Совета Фонда не принято, а также в случае, указанном в п. 7.5, заседание Совета Фонда может быть созвано членом Совета Фонда, требующим его проведения.
- 7.8. Исполнительный директор или иное лицо, созывающее собрание Совета Фонда в соответствии с п. 7.7, извещает членов Совета Фонда о проведении заседания и о повестке дня с приложением всех необходимых материалов и информации для ознакомления посредством их направления каждому члену Совета Фонда по электронной почте не позднее чем за 2 (два) рабочих дня до даты проведения заседания. Во избежание сомнений, Исполнительный директор извещает членов Совета Фонда о проведении заседания и о повестке дня незамедлительно после назначения даты для проведения собрания в соответствии с п. 7.7.
- 7.9. Члены Совета Фонда вправе вносить предложения по включению в повестку дня дополнительных вопросов в течение 1 (одного) рабочего дня после их уведомления о проведении заседания Совета Фонда. В случае включения в повестку дня дополнительных вопросов, информация об этом доводится до сведения членов Совета Фонда не позднее 1 (одного) рабочего дня до даты проведения заседания.

- 7.10. In case of violation of the procedure established by this Charter for convening and preparing a meeting of the Board, such meeting shall nonetheless be recognized as valid if it was attended by all incumbent members of the Board.
- 7.11. A meeting of the Board is competent (has a quorum) if attended by a majority of the incumbent members of the Board, including at least one Director A. If the agenda of the meeting includes issues the approval of which requires unanimous vote, or vote by a qualified majority, or by a specific number of the members of the Board, the meeting of the Board will be competent to consider such issues provided that the meeting of the Board whose number is sufficient to pass a relevant decision.
- 7.12. Each of the members of the Board participating in the meeting shall have one vote, other than (i) exercising the vote of another member of the Board in accordance with paragraph 6.8 and (ii) in the circumstances when a special procedure for decision-making on certain matters applies as set forth in Schedule 4.
- 7.13. In the absence of a quorum, the Executive Director, or the person who has convened the Board meeting, shall announce the convocation of another meeting of the Board instead of the one that failed, and such new meeting shall be with the same agenda and shall be held no later than two (2) business days from the date of the failed meeting of the Board. The quorum of the meeting called instead of the failed meeting shall be as required under paragraph 7.11, provided however that no participation of at least one Director A is required, as long as the votes of Directors A shall not be required for adopting a decision on any matter included on such meeting's agenda.

- 7.10. В случае нарушения установленного настоящим Уставом порядка созыва и подготовки заседания Совета Фонда, такое заседание признается правомочным, если в нем участвуют все действующие члены Совета Фонда.
- 7.11. Заседание Совета Фонда правомочно (имеет кворум), если в нем принимает участие большинство действующих членов Совета Фонда, включая как минимум одного Директора А. В случае включения в повестку дня вопросов, принятие решения по которым требует единогласия, голосования квалифицированного большинства или определенного количества членов Совета Фонда, заседание Совета Фонда правомочно рассматривать такие вопросы при условии участия в заседании достаточного для принятия решения количества членов Совета Фонда.
- 7.12. Каждый член Совета Фонда, участвующий в заседании, обладает одним голосом, за исключением (i) случая осуществления права голоса другого члена Совета Фонда в соответствии с п. 6.8 и (ii) случаев специального порядка принятия решений по отдельным вопросам компетенции в соответствии с Приложением 4.
- 7.13. В случае отсутствия кворума Исполнительный директор или лицо, созвавшее заседание Совета Фонда, объявляет о созыве повторного заседания Совета Фонда взамен несостоявшегося с такой же повесткой дня, и такое заседание должно быть проведено не позднее 2 (двух) рабочих дней с даты несостоявшегося заседания Совета Фонда. К повторному заседанию применяются требования о кворуме, указанные в п. 7.11 с тем условием, однако, что не требуется участие как минимум одного Директора А, если голоса Директоров А не требуются для принятия решения ни по одному из вопросов повестки дня.

- 7.14. A meeting of the Board shall be opened by the Executive Director or a member of the Board convening the meeting, who shall hold the election of the chair from among the members of the Board attending the meeting. The chair is elected by a simple majority of the members of the Board participating in the meeting. The chair shall conduct the meeting, count the votes and adjourn the meeting. The Board member that has requested the meeting convocation shall act as the Secretary, and if such Board member is not participating in such Board meeting, or has been elected as the chair, then the Foundation Secretary shall act as the Secretary. If the Foundation does not have a position of Foundation Secretary, the Executive Director shall act as the Secretary. For the purposes of certification of the Board decisions referred in paragraph 7.20, the meeting should be attended by a Notary.
- 7.15. Other than envisaged by paragraph 7.20, 7.15. for the purposes of determining the quorum and voting results on the agenda of the Board meeting, the votes of the following Board members shall be considered in addition to those of Board members participating in person:
- 7.15.1. vote of the Board member participating in the meeting via video conferencing or similar means of communication allowing all the participants in the meeting of the Board to see and hear all other members of the Board who participate in person, provided that such option of participation was notified in the notice of the meeting in accordance with paragraph 7.8;
- 7.15.2. vote of the Board member who has authorized another Board member of the same class to exercise his/her voting rights at such Board meeting in

7.14. Заседание Совета Фонда открывается Исполнительным директором или членом Совета Фонда, созвавшим заседание, который проводит выборы председательствующего лица из числа принимающих участие в заседании членов Совета Фонда. Председательствующее лицо избирается простым большинством участвующих в заседании членов Совета Фонда.

Председательствующее лицо ведет заседание, подводит подсчет голосов и объявляет его закрытым. Функции Секретаря исполняет член Совета Фонда, по требованию которого созвано данное заседание, а в случае, если такой член Совета Фонда не принимает участие в заседании или избран председательствующим, то Секретарь Фонда. В случае если в Фонде отсутствует позиция Секретаря Фонда, функции Секретаря исполняет Исполнительный директор. Для целей удостоверения решений Совета Фонда в случаях, предусмотренных в п. 7.20, на заседаниях должен присутствовать Нотариус.

- 15. За исключением случаев, указанных в п. 7.20, для целей определения наличия кворума и подведения результатов голосования по вопросам повестки заседания Совета Фонда помимо голосов членов Совета Фонда, присутствующих лично, также учитываются:
- 7.15.1. голос члена Совета Фонда, участвующий в заседании с использованием видеоконференции, или аналогичных средств связи, позволяющих всем участникам заседания Совета Фонда видеть и слышать всех других членов Совета Фонда, принимающих участие лично, при условии, что такой способ участия был определен в извещении о проведения заседания в соответствии с п. 7.8;
- 7.15.2. голос члена Совета Фонда, который уполномочил другого члена Совета Фонда того же класса осуществлять его/ее право голоса на таком

accordance with paragraph 6.8.5, provided that (i) not more than three of Directors B may authorize another Director B to exercise their voting rights at such meeting under a power of attorney, and provided that (ii) each member of the Board participating in the meeting in person may be authorized to cast votes under a power of attorney from not more than two other members of the Board of the same class; and

- 7.15.3. vote of an absent Board member who has not authorized another Board member to exercise the voting rights in accordance with 7.15.2, but who has submitted a completed written ballot on the meeting agenda and specified his/her vote on each agenda item. Such written ballot shall be taken into account for the purposes of the quorum and voting on the agenda if all of the below conditions are met:
 - (a) the authenticity of the Board member's signature on the written ballot on the agenda has been certified by a notary; and
 - (b) the original written ballot on the agenda executed by such Board member has been submitted before the end of the registration of the members of the Board for the participation in the meeting specified in the convocation notice; and
 - (c) the Board member has expressly specified how his/her vote shall be casted on each item of the agenda (i.e. "in favor", "against", "abstained").
- 7.16. No later than two (2) business days after the meeting of the Board, minutes of the meeting must be drawn up specifying

- заседании в порядке, указанном в п. 6.8.5, при условии, что (i) не более чем три Директора В могут уполномочить другого действующего Директора В на осуществление права голоса в таком заседании по доверенности, и при этом (ii) каждый член Совета Фонда, участвующий в заседании Совета Фонда лично, может быть уполномочен на распоряжаться по доверенности правом голоса не более, чем двух членов Совета Фонда того же класса; и
- 7.15.3. голос отсутствующего члена Совета Фонда, не уполномочившего другого члена Совета Фонда осуществлять право голоса в порядке, указанном в п. 7.15.2, который направил заполненный бюллетень по вопросам повестки дня заседания Совета Фонда с указанием варианта голосования по каждому вопросу. Такой бюллетень должен учитываться для определения кворума и результатов голосования, если соблюдены одновременно все следующие условия:
 - (а) подлинность подписи члена Совета Фонда, направившего бюллетень по вопросам повестки дня, удостоверена нотариально;
 - (б) оригинал бюллетеня такого члена Совета Фонда по вопросам повестки дня заседания предоставлен не позднее времени окончания регистрации членов Совета Фонда для участия в заседании, указанного в извещении о созыве заседания; и
 - (в) член Совета Фонда однозначно определил свою позицию по каждому вопросу повестки дня, указав конкретный вариант голосования («за», «против» или «воздержался»).
- 7.16. Не позднее 2 (двух) рабочих дней после проведения заседания Совета Фонда составляется протокол с

the agenda of the meeting, the decisions made, and the voting results. The minutes of the meeting of the Board must be executed by the chair of the meeting and the Secretary, as well as by all members of the Board who participated in the meeting in person, unless otherwise required by this Charter.

The minutes of the Board meeting shall append all written ballots and powers of attorney, representing the votes that have been taken into account for the purposes of the quorum and voting on the agenda of the Board meeting in accordance with paragraph 7.15.

- 7.17. Copies and extracts from the minutes of the meeting of the Board shall be signed by the Executive Director.
- 7.18. For the purposes of this Charter, unless otherwise specifically required, any notices by the Board members to the Foundation and other Board members, as well as from the Foundation to the Board members in connection with convocation and preparing to the Board meetings may be sent by email or other electronic means of communication with a copy by regular mail or courier delivery at the address of the addressee.
- 7.19. In the event the Board passes a decision on any matters listed in item 11 of Schedule 4 such decision shall also authorize one of Directors B to participate in the general participants' meeting of Yandex LLC and exercise Foundation's voting right in accordance with the decision of the Board.
- 7.20. The Board is entitled to consider the matters specified in item 11(b) and 11(e) of <u>Schedule 4</u> hereto, in each case only subject to satisfaction of all conditions requires for the occurrence of a Special Corporate Situation or a

указанием повестки дня, принятых решений и результатов голосования по ним. Протокол заседания Совета Фонда подписывается председательствующим лицом, Секретарем, а также всеми членами Совета Фонда, участвовавшими в заседании лично, если иное не предусмотрено Уставом.

К протоколу заседания Совета Фонда приобщаются все бюллетени для голосования и доверенности, голоса по которым учитываются для целей определения наличия кворума и подведения результатов голосования в соответствии с п. 7.15.

- 7.17. Копии и выписки из протокола заседания Совета Фонда подписывает Исполнительный директор.
- 7.18. Для целей настоящего Устава, если специально не указано иное, все уведомления членов Совета Фонда в адрес Фонда и других членов Совета Фонда, а также от имени Фонда членам Совета Фонда в связи с созывом и подготовкой к проведению заседания Совета Фонда могут направляться посредством электронной почты и иными средствами электронной связи с направлением копии такого уведомления по почте или курьерской доставкой по адресу местонахождения адресата.
- 7.19. В случае принятия Советом Фонда решения по вопросам, перечисленным в п. 11 Приложения 4, Совет Фонда обязан указанным решением также уполномочить одного из Директоров В участвовать в общем собрании участников ООО Яндекс от имени Фонда и осуществлять право голоса Фонда в соответствии с принятым Советом Фондом решением.
- 7.20. Совет Фонда правомочен рассматривать вопросы, указанные в п. 11(б) и 11(д) <u>Приложения 4</u> к Уставу, в каждом случае только в случае выполнения всех указанных в <u>Приложении 5</u> условий для

Special Situation specified in <u>Schedule 5</u>, as applicable, and subject to compliance with the following procedures for making and certifying the decision:

- 7.20.1. the relevant Board meeting is held in person (and provisions of paragraph 7.15 do not apply); and
- 7.20.2. if any of the incumbent Directors B whose vote is required for the quorum is not able to attend the meeting of the Board in person, such Director B may authorize another incumbent Director B to exercise his/her voting rights on the agenda of such Board meeting in accordance with paragraph 6.8.4, provided that (i) not more than three of Directors B may authorize another Director B to exercise their voting rights at such meeting under a power of attorney, and provided that (ii) each Director B participating in the meeting in person may be authorized to cast votes under a power of attorney from not more than two Directors B; and

7.20.3. the fact of taking the decision by the Board on items 11(b) and 11(e) of Schedule 4 hereto shall be certified by a notary in accordance with the procedure set out in Art. 103.10 of Fundamentals of Legislation of the Russian Federation on Notarial Activities.

8. THE EXECUTIVE DIRECTOR

- 8.1. The Executive Director is responsible for day-to-day management of the activities of the Foundation and acts on behalf of the Foundation without a power of attorney.
- 8.2. The Executive Director is elected by the decision of the Board from among

- наступления Особой корпоративной ситуации и Особой ситуации, соответственно, и с соблюдением следующего порядка принятия и удостоверения факта принятия решений:
- 7.20.1. соответствующее заседание Совета Фонда проводится в очной форме (положения п. 7.15 не применяются); и
- 7.20.2. если какой-либо из действующих Директоров В, чей голос необходим для достижения кворума для принятия указанного решения, не может участвовать в заседании Совета Фонда лично, такой Директор В вправе уполномочить другого действующего Директора В на осуществление права голоса по вопросам повестки дня такого заседания путем предоставления ему доверенности в нотариальной форме в порядке, указанном в п. 6.8.4, при условии, что (i) не более чем три Директора В могут уполномочить другого действующего Директора В на осуществление права голоса в таком заселании по доверенности, и при этом (іі) каждый Директор В, участвующий в заседании Совета Фонда лично, может быть уполномочен распоряжаться по доверенности правом голоса не более, чем двух Директоров В; и
- 7.20.3. факт принятия решения Совета Фонда по п. 11(б) и 11(д) Приложения 4 подлежит удостоверению Нотариусом в порядке, предусмотренном ст. 103.10 Основ законодательства Российской Федерации о нотариате.

8. ИСПОЛНИТЕЛЬНЫЙ ДИРЕКТОР

- 8.1. Исполнительный директор осуществляет текущее руководство деятельностью Фонда и действует от имени Фонда без доверенности.
- 8.2. Исполнительный директор избирается решением Совета Фонда

Directors A for an indefinite term. The initial Executive Director acting from the date of this version of the Charter is elected by the Founder upon the decision of the approval of this version of the Charter. The terms of the employment contract of the initial Executive Director are determined by the Founder.

- из числа Директоров А на неопределенный срок. Первоначальный Исполнительный директор, действующий с даты принятия настоящей редакции Устава, избирается Учредителем при утверждении настоящей редакции Устава. Условия трудового договора первоначального Исполнительного директора определяются Учредителем.
- 8.3. In performing his activities, the Executive Director is reporting to the Board and is responsible for ensuring the fulfillment of decisions of the Board.
- 8.3. В своей деятельности Исполнительный директор подотчетен Совету Фонда и организует выполнение решений Совета Фонда.
- 8.4. The Executive Director:
- 8.4. Исполнительный директор:
- 8.4.1. acts on behalf of the Foundation without a power of attorney, including representing the Foundation in dealings with third parties within the scope of powers stipulated in the Charter;
- 8.4.1. действует от имени Фонда без доверенности, в том числе представляет его интересы в отношениях с третьими лицами в пределах, установленных Уставом;
- 8.4.2. prepares and submits for the approval by the Board annual financial plans (budgets) of the Foundation;
- 8.4.2. готовит и выносит на одобрение Советом Фонда финансовый план (бюджет) Фонда на год;
- 8.4.3. enters into transactions and agreements on behalf of the Foundation subject to applicable restrictions hereunder (including restrictions set out in paragraphs 4.5 and 8.5);
- 8.4.3. заключает сделки и договоры от имени Фонда с соблюдением установленных Уставом ограничений (в том числе указанных в пп. 4.5 и 8.5);
- 8.4.4. convenes and opens meetings of the Board;
- 8.4.4. созывает и открывает заседания Совета Фонда;
- 8.4.5. approves internal labor regulations and ensures compliance therewith;
- 8.4.5. утверждает правила внутреннего трудового распорядка и обеспечивает их соблюдение;
- 8.4.6. issues orders on the appointment of employees of the Foundation, on their transfers and dismissals, applies incentive and disciplinary measures to them within the scope provided for in the financial plan (budget) of the Foundation;
- 8.4.6. издает приказы о назначении на должности работников Фонда, об их переводе и увольнении, применяет к ним меры поощрения и дисциплинарной ответственности в рамках финансового плана (бюджета) Фонда;
- 8.4.7. opens and closes settlement and other accounts of the Foundation in banking and lending institutions in Russian and foreign currency;
- 8.4.7. открывает и закрывает расчетные и иные счета Фонда в банковских и кредитных организациях в российской и иностранной валюте;

- 8.4.8. issues orders, instructions, directives and other acts on matters falling within his scope of competence, that are binding on the employees of the Foundation:
- 8.4.9. represents and protects the rights and legitimate interests of the Foundation in relations with the bodies of legislative, executive and judicial power, local governments, as well as in relations with legal entities and individuals;
- 8.4.10. annually reports on the work done to the Board and the Founder;
- 8.4.11. organizes the accounting and reporting work;
- 8.4.12. ensures compliance with the provisions of this Charter, the main areas of the Foundation's activities and other documents of the Foundation;
- 8.4.13. makes decisions on other issues related to the activities of the Foundation which were not referred by this Charter and Applicable Law to the competence of the other management bodies of the Foundation.
- 8.5. The Executive Director is not entitled to exercise the following rights at his/her sole discretion, without first receiving an approval of the Board:
- 8.5.1. exercise any rights attaching to the Priority Share or Special Interest, as well as any rights of participation in other commercial entities,
- 8.5.2. enter into gratuitous transactions 8.5.2. whereby the Foundation is to alienate any of its assets;
- 8.5.3. enter into, amend and terminate any agreements on voting, exercising corporate rights, issuing instructions to

- 8.4.8. издает приказы, указания, распоряжения и другие акты по вопросам, входящим в его компетенцию, обязательные для работников Фонда;
- 8.4.9. представляет и защищает права и законные интересы Фонда в органах законодательной, исполнительной и судебной власти, местного самоуправления, а также в отношениях с юридическими и физическими лицами;
- 8.4.10. ежегодно отчитывается о проделанной работе перед Советом Фонда и Учредителем;
- 8.4.11. организует ведение бухгалтерского учета и отчетности;
- 8.4.12. обеспечивает выполнение положений настоящего Устава, основных направлений деятельности Фонда и иных документов Фонда;
- 8.4.13. принимает решения по иным вопросам, связанным с деятельностью Фонда, решение которых не отнесено настоящим Уставом и Действующим законодательством к компетенции других органов Фонда.
- 8.5. Исполнительный директор не вправе по своему усмотрению реализовывать следующие полномочия без получения предварительного согласия Совета Фонда:
- 8.5.1. осуществлять какие-либо права, предоставленные Специальной акцией или Специальной долей, а равно иные права участия в какихлибо хозяйственных обществах;
- 8.5.2. заключать безвозмездные сделки, по которым Фонд отчуждает какое-либо имущество;
- 8.5.3. заключать, изменять и прекращать какие-либо соглашения о порядке голосования, осуществления корпоративных прав, выдачи

the members of the governing bodies of other entities:

- 8.5.4. enter into any transactions with any third parties in excess of the total amount of one (1) million US dollars in an individual financial year, or, with respect to transactions specified in paragraph 10.6, in excess of the total amount of six million (6,000,000) Rubles in any given financial year (or an equivalent amount in other currency, in each case).
- The Executive Director may be 8.6. removed from his/her position before the expiration of his/her term of office by decision of the Board, e.g., due to the Director ceasing to be a member of the Board, or may resign at his/her discretion. For the avoidance of doubt, the termination of the Executive Director's powers as such does not trigger termination of his/her powers as the relevant member of the Board.

THE RIGHTS OF THE FOUNDER 9.

- 9.1. The Founder exercises its rights in relation to the Foundation as provided for by the Applicable Law and this Charter.
- 9.2. The management bodies of the Foundation may not pass any of the following decisions without a prior approval by the Founder:
- 9.2.1. amendment of the Charter;
- 9.2.2. liquidation of the Foundation;
- 9.2.3. approval of the financial plan (budget) of the Foundation for the financial year and amendments thereto;
- approval of the Foundation's staffing table, including the amount of remuneration:

инструкций членам органов управления других организаций;

- 8.5.4. заключать какие-либо сделки с любыми третьими лицами на общую сумму свыше 1 (одного) миллиона долларов США в течение финансового года, а в отношении сделок, указанных в п. 10.6, - на общую сумму свыше 6 000 000 (шести миллионов) рублей в течение финансового года (или эквивалента в другой валюте, в каждом случае).
- 8.6. Полномочия Исполнительного директора могут быть прекращены досрочно по решению Совета Фонда, в том числе в связи с прекращением у лица, занимающего должность Исполнительного директора, статуса члена Совета Фонда, а также в случае добровольного сложения с себя полномочий. Прекращение полномочий Исполнительного директора само по себе не влечет прекращение полномочий члена Совета Фонда.

9. ПРАВА УЧРЕДИТЕЛЯ

- 9.1. Учредитель реализует свои права в отношении Фонда способами, предусмотренными Действующим законодательством и Уставом.
- 9.2. Не допускается принятие органами управления Фондом следующих решений без предварительного согласия Учредителя:
- 9.2.1. о внесении изменений и дополнений в Устав:
- 9.2.2. о ликвидации Фонда;
- 9.2.3. об утверждении финансового плана (бюджета) Фонда на очередной финансовый год или внесении изменений в него:
- 9.2.4. об утверждении штатного расписания, Фонда, включая размеры оплаты труда;

- 9.2.5. issuance of powers of attorney on behalf of the Foundation.
- 9.3. To the extent any approval or consent is required from the Founder in accordance with this Charter, such approval or consent should be specific, made in writing and executed by the CEO or another authorized representative of the Founder, whose signature must be certified by a notary (practicing at the location where the consent is being executed).

10. CONFLICT OF INTERESTS

- 10.1. A member of the Foundation's management body is considered to be an interested party if he/she, his/her spouse, parents, children, half- and full brothers and sisters, adoptive parents and/or adopted children and/or persons under their control (controlled entities), any other relatives on any side or persons residing together with the member of the Foundation's management body:
- 10.1.1. are a party, beneficiary, intermediary or representative in the transaction of the Foundation;
- 10.1.2. are the Controlling person of a legal entity that is a party, beneficiary, intermediary or representative in the transaction of the Foundation;
- 10.1.3. hold positions in the management bodies of a legal entity that is a party, beneficiary, intermediary or representative in a transaction with the Foundation, or positions in the management bodies of the managing organization of that legal entity.
- 10.2. Interested parties shall, within 3 (three) days from the day they learned or should have learned about the circumstances due to which they may be recognized as interested in the decisions

- 9.2.5. о выдаче доверенностей от имени Фонда.
- 9.3. В случае если настоящим Уставом требуется получение какого-либо согласия или одобрения Учредителя, такое согласие или одобрение должно быть явно выраженным, оформленным письменно и подписано СЕО или иным лицом, уполномоченным Учредителем, чья подпись должна быть заверена нотариусом (осуществляющим деятельность по месту оформления согласия).

10. КОНФЛИКТ ИНТЕРЕСОВ

- 10.1. Заинтересованным признается член органа управления Фонда в том случае, если он, его супруг(а), родители, дети, полнородные и неполнородные братья и сестры, усыновители и усыновленные и (или) подконтрольные им лица (подконтрольные организации), любые иные родственники по любой линии или лица, проживающие совместно с членом органа управления Фонда:
- 10.1.1. являются стороной, выгодоприобретателем, посредником или представителем в сделке Фонда;
- 10.1.2. являются Контролирующим лицом юридического лица, являющегося стороной, выгодоприобретателем, посредником или представителем в сделке Фонда;
- 10.1.3. занимают должности в органах управления юридического лица, являющегося стороной, выгодоприобретателем, посредником или представителем в сделке с Фондом, а также должности в органах управления управляющей организации такого юридического лица.
- 10.2. Заинтересованные лица обязаны в течение 3 (трех) дней со дня, когда они узнали или должны были узнать об обстоятельствах, в силу которых они могут быть признаны

of the management bodies and transactions of the Foundation, notify the Foundation and the Founder of the persons fitting the description provided in paragraph 10.1, and the proposed transactions and decisions of the management bodies of the Foundation of which they are aware and in which they may be recognized as being interested.

- 10.3. In the event of a change in any of the above circumstances, interested parties are obliged to notify the Foundation and the Founder of the change within three (3) days from the day they learned or should have learned about the change.
- 10.4. The Foundation shall bring the information contained in the notification received by it pursuant to paragraphs 10.2 and 10.3 of this section to the notice of the management bodies of the Foundation.
- 10.5. Interested party transaction of the Foundation requires prior approval by the Board, unless paragraph 10.6 applies.
- 10.6. No prior approval is required under paragraphs 8.5.4 and 10.5 for any transactions by the Foundation, in which Director A (Directors A) is interested, as long as the total value of such transactions does not exceed 6,000,000 (six million) Rubles during an individual financial year (or equivalent in other currency).
- 10.7. A decision to approve an interested party transaction shall be made by a majority of votes of the members of the Board. If a member of the Board is recognized as an interested party in relation to a decision or a transaction, his vote shall be disregarded when determining the availability of the quorum and the results of the voting on the relevant agenda item. If all members of the Board are interested in a

заинтересованными в решениях органов управления и сделках Фонда, уведомить Фонд и Учредителя о лицах, в отношении которых имеются обстоятельства, перечисленные в пункте 10.1, и об известных им совершаемых или предполагаемых сделках и решениях органов управления Фонда, в которых они могут быть признаны заинтересованными.

- 10.3. В случае изменения каких-либо указанных обстоятельств заинтересованные лица обязаны уведомить Фонд и Учредителя об их изменении в течение 3 (трех) дней со дня, когда они узнали или должны были узнать о таком изменении.
- 10.4. Фонд доводит информацию, содержащуюся в полученном им уведомлении, предусмотренном пунктами 10.2 и 10.3 настоящей статьи, до сведения органов управления Фонда.
- 10.5. Сделка Фонда, в которой имеется заинтересованность, требует обязательного предварительного одобрения Советом Фонда за исключением случаев, указанных в п. 10.6.
- 10.6. Не требуют одобрения в соответствии с п. 8.5.4 и п. 10.5 какие-либо сделки Фонда, в которых имеется заинтересованность Директора А (Директоров А), если общая стоимость таких сделок не превышает 6 000 000 (шесть миллионов) рублей в течение одного финансового года (или эквивалент в другой валюте).
- 10.7. Решение об одобрении сделки, в которой имеется заинтересованность, принимается большинством голосов незаинтересованных членов Совета Фонда. В случае заинтересованности члена Совета Фонда в принятии такого решения или совершении такой сделки, его голос не учитывается для целей определения кворума и результатов голосования по соответствующему вопросу

transaction, a decision to approve such transaction shall be made in the ordinary manner (if it requires approval in accordance with the Charter on other grounds) without following the voting procedure established in this paragraph. повестки. В случае заинтересованности в сделке всех членов Совета Фонда решение об ее одобрении принимается в обычном порядке (если ее одобрение предусмотрено Уставом по иным основаниям) без применения порядка голосования, установленного в настоящем пункте.

- 10.8. For avoidance of doubt provisions of this section 10 shall not be interpreted to restrict the voting powers of Directors A in respect of all matters listed in items 10 and 11 of Schedule 4 regardless of whether Directors A may be viewed as interested in the transactions and decisions proposed for considerations by the Board.
- 10.8. Во избежание сомнений, положения настоящего раздела 10 ни в коей мере не ограничивают право голоса Директоров А по всем вопросам, указанным в пп. 10-11 Приложения 4 вне зависимости от того, могут ли они быть признаны заинтересованными в совершении сделок и действий, которые выносятся на одобрение Совета Фонда.

11. MAKING AMENDMENTS TO THE CHARTER

11. ПОРЯДОК ВНЕСЕНИЯ ИЗМЕНЕНИЙ В УСТАВ

- 11.1. The Board is not authorized to approve any changes and additions to this Charter in the following cases:
- 11.1. Совет Фонда не вправе принять решение о внесении изменений и дополнений в Устав в следующих случаях:
- 11.1.1. The Board has not obtained a prior approval by the Founder to such changes and additions;
- 11.1.1. Совет Фонда не получил предварительное согласие Учредителя на внесение таких изменений и дополнений;
- 11.1.2. there is a vacancy on the Board.
- 11.1.2. Совет Фонда не полностью сформирован.
- 11.2. The state registration of changes and additions to this Charter is made in the manner prescribed by Applicable Law.
- 11.2. Государственная регистрация изменений и дополнений к настоящему Уставу осуществляется в порядке, установленном Действующим законодательством.
- 11.3. Changes and additions to this Charter become effective from the time they are recorded in the unified state register of legal entities.
- 11.3. Изменения и дополнения к настоящему Уставу вступают в силу с момента внесения соответствующей записи в единый государственный реестр юридических лиц.

12. LIQUIDATION OF THE FOUNDATION

12. ЛИКВИДАЦИЯ ФОНДА

12.1. The Foundation may be liquidated by unanimous decision of the Board, with

12.1. Фонд может быть ликвидирован по единогласному решению Совета

the prior written approval of the Founder, in case the object of the Foundation has been achieved or it has become apparent that the object cannot be achieved or amended.

- 12.2. The Foundation may be liquidated by a court decision in cases provided for by the Law.
- 12.3. Upon liquidation, the assets of the Foundation remaining after payment of creditor claims, including the assets referred to in paragraph 3.2.1 of the Charter, shall be transferred to the Founder.
- 12.4. The reorganization of the Foundation is not allowed.

13. MISCELLANEOUS

- 13.1. Other than set out in paragraph 7.18, all notices, requests and communications envisaged by this Charter shall be delivered by personal delivery or prepaid urgent delivery by a recognized international courier service.
- 13.2. Any notice to be given under this Charter, provided it has been duly addressed, shall be deemed to have been received:
- 13.2.1. if personally delivered, at the time of delivery as specified by the acceptance stamp by a representative of the receiver;
- 13.2.2. if sent by pre-paid urgent delivery by an international express delivery service (such as UPS, DHL, FedEx or similar), three (3) business days after the date of posting to the relevant address or upon the receipt of a delivery report of a courier or a report of an international express delivery service (UPS, DHL, FedEx or similar) regarding the recipient's refusal to accept the message

Фонда при наличии предварительного письменного согласия Учредителя в случае достижения уставной цели Фонда либо очевидной невозможности достижения уставной цели и ее изменения.

- 12.2. Фонд может быть ликвидирован по решению суда в случаях, предусмотренных Законом.
- 12.3. В случае ликвидации имущество Фонда, оставшееся после расчетов с кредиторами, включая имущество, указанное в п. 3.2.1 Устава, подлежит передаче Учредителю.
- 12.4. Реорганизация Фонда не допускается.

13. ПРОЧИЕ ПОЛОЖЕНИЯ

- 13.1. За исключением случаев, указанных в п. 7.18, все уведомления, требования и сообщения, предусмотренные настоящим Уставом, должны направляться посредством личной доставки или посредством срочного отправления с предоплатой услуг через признанную международную курьерскую службу.
- 13.2. Какое-либо уведомление, предусмотренное настоящим Уставом, которое было надлежащим образом адресовано, будет считаться полученным:
- 13.2.1. в случае доставки курьером: в момент доставки, указанный путем отметки о получении уведомления представителем получающего лица;
- 13.2.2. в случае передачи срочным отправлением с предоплатой услуг через международную службу экспресс доставки (UPS, DHL, FedEx или аналогичные): через 3 (три) рабочих дня после даты отправления на соответствующий адрес либо в момент получения отчета курьера или отчета международной службы экспресс доставки (UPS, DHL, FedEx или аналогичные) об отказе

- or inability to deliver to the specified address; and
- получателя принять отправление или невозможности доставки по указанному адресу; и
- 13.2.3. in case of email delivery: on the sent day if sent prior to 18.00, or on the following calendar date if sent after 18.00, Moscow time in each case.
- 13.2.3. в случае отправки по электронной почте: в день отправки (в случае направления до 18.00), на следующий календарный день после отправки (в случае направления после 18.00), в каждом случае по московскому времени.
- 13.3. This Charter is made in English and Russian languages, with both versions intended to be identical in meaning and of equal legal force; in the event of any discrepancy the Russian text shall prevail.
- 13.3. Устав составлен на английском и русском языках, при этом оба варианта должны быть идентичными по смыслу и иметь равную юридическую силу; в случае несоответствия текст на русском языке имеет преимущественную силу.

SCHEDULE 1. REQUIREMENTS

Any person who is a member of the management bodies of the Foundation (as well as each candidate for the position in the management bodies of the Foundation), must meet the following requirements:

- (1) may not be an employee of a state corporation or a company controlled by the state (or by any state agency).
- (2) may not be currently and should not have been during the last 2 full calendar years a person appointed by a political party or be a member of the governance bodies of a political party, or a person holding a position as a civil servant, a member / employee of any government body, or a member of an elected body of power, or hold any other public office.
- (3) no criminal record; not subject to disqualification under the Code of Administrative Offenses of the Russian Federation; not subject to any administrative penalty for any offense listed in chapter 15 of the Code of Administrative Offenses of the Russian Federation.
- (4) may not be a person with whom the Foundation or Yandex Group may not be allowed to have dealings due to restrictions imposed by law applicable to the Foundation or any person within Yandex Group.
- (5) may not have currently and should not have had during the last two full calendar years a commercial conflict of interest with Yandex Group. For the purpose of this provision:
 - A commercial conflict of interest is defined as the existence of a commercial relationship between a

ПРИЛОЖЕНИЕ 1. ТРЕБОВАНИЯ

Каждое лицо, входящее в состав органов управления Фонда (а равно кандидат на должность в органах управления Фонда) должно соответствовать следующим требованиям:

- (1) Не должно быть сотрудником государственной корпорации или компании, находящейся под контролем государства (или какого-либо государственного органа).
- (2) Не должно быть на текущий момент и в течение последних 2-ух полных календарных лет лицом, назначенным политической партией или состоящим в органах управления политической партией, лицом, занимающим должность государственного служащего, членом/сотрудником какого-либо органа государственной власти, членом выборного органа власти, занимать государственную должность.
- (3) Не должно иметь судимость, являться дисквалифицированным лицом в соответствии с Кодексом Российской Федерации об административных правонарушениях, считаться подвергнутым административному наказанию за какое-либо административное правонарушение, указанное в главе 15 Кодекса Российской Федерации об административных правонарушениях.
- (4) Не должно быть лицом, с которыми Фонду или Группе Яндекс может быть запрещено сотрудничать в силу ограничений, наложенных правом, применимым к Фонду или любому лицу Группы Яндекс.
- (5) Не должно на текущий момент и в течение последних 2-ух полных календарных лет иметь коммерческий конфликт интересов с Группой Яндекс. При этом:
 - Коммерческий конфликт интересов определяется как наличие коммерческих отношений

business that competes with the business of Yandex Group, and:

- the member of the management bodies (candidate to the position of in the management bodies) of the Foundation;
- his/her close relatives (including his/her spouse, parents, spouse's parents, children, siblings) or any person sharing the person's household:
- legal entities that are, directly or indirectly, Controlled Persons of such persons or his/her close relatives;
- o entities in which the candidate or his/her close relatives have a shareholding of more than 1% of a listed company or 3% of a non-public company;
- Commercial relationships include the following direct or indirect relationships with a business that competes with the business of Yandex Group:
 - labor relations, including past labor relations:
 - membership in the board of directors, including past membership;
 - acting as a consultant;
 - o ownership of shares (more than 1% in a public company or 3% in a non-public company).
- A business that competes with business of Yandex Group - a business that operates in any sector/field of activity (save for the areas/lines of activities in academic

между бизнесом, который конкурирует с бизнесом Группы Яндекс, и:

- членом органа управления (кандидатом на должность члена органов управления)
 Фонда;
- о его (ее) близкими родственниками (включая, его(ее) супругу/супруга, родителей, родителей супруга/супруги, детей, братьев или сестер) или лицами, проживающими совместно с таким лицом;
- юридическими лицами, являющимися прямо или косвенно Контролируемыми лицами такого лица или его (ее) близких родственников;
- о компаниями, в которых лицо или его (ее) близкие родственники владеют пакетом акций более 1% в публичной компании или 3% в непубличной компании.
- Коммерческие отношения включают в себя следующие прямые или косвенные отношения с бизнесом, который конкурирует с бизнесом Группы Яндекс:
 - о трудовые отношения, в том числе в прошлом;
 - членство в совете директоров, в том числе в прошлом;
 - о оказание услуг в качестве консультанта;
 - владение акциями (более 1% в публичной компании или 3% в непубличной компании).
- Бизнес, который конкурирует с бизнесом Группы Яндекс бизнес, который работает в любом секторе/сфере деятельности (за исключением секторов/сфер

science, education and not-for-profit medicine) in which Yandex Group's revenue is more than 1% of the total revenue of Yandex Group based on the consolidated financial statements prepared under generally accepted accounting principles in the US (*US GAAP*) for the six-month period ending on the last balance sheet date, as determined in accordance with US GAAP.

академической науки, образования и некоммерческой медицины), в котором выручка Группы Яндекс составляет более 1% в общей выручке Группы Яндекс по консолидированной отчетности Группы Яндекс, подготовленной по общепринятым стандартам бухгалтерской отчетности США (US GAAP), за шестимесячный период, оканчивающийся в последнюю дату баланса, как она определяется в соответствии с US GAAP.

SCHEDULE 2. QUESTIONNAIRE

I, [full name], hereby certify that I am aware of the Requirements as set out in the Foundation Charter and make the below statements as of the date hereof:

I am not currently an employee of a state corporation or companies controlled by the state:	1.	Я не являюсь сотрудником государственной корпорации или компании, находящейся под контролем государства:
☐ confirm / ☐ do not confirm, see details below:		подтверждаю / пе подтверждаю соответствующие сведения указаны ниже:
I am not currently and within the preceding	2.	Я не являюсь на текущий момент и в
two (2) full calendar years have not been a political appointee or member of governing bodies of a political party, a government official or a member or employee of any state apparatus, a member of parliament, or a political office-holder:	2.	течение последних 2 (двух) полных календарных лет не являлся лицом, назначенным политической партией или состоящим в органах управления политической партией, лицом, занимающим должность государственного служащего, членом/сотрудником какого-либо органа государственной власти, членом выборного органа власти и не занимал государственную должность:
☐ confirm / ☐ do not confirm, see details below:		☐ подтверждаю / ☐ не подтверждаю соответствующие сведения указаны ниже:
I do not have criminal record; am not subject to disqualification under the Code of Administrative Offenses of the Russian Federation, am not subject to any administrative penalty for any offense listed in chapter 15 of the Code of Administrative Offenses of the Russian Federation:	3.	Я не имею судимостей, не являюсь дисквалифицированным лицом в соответствии с Кодексом Российской Федерации об административных правонарушениях, не считаюсь подвергнутым административному наказанию за какое-либо административное правонарушение, указанное в главе 15 Кодекса Российской Федерации об административных правонарушениях:
☐ confirm / ☐ do not confirm, see details below:		подтверждаю / пе подтверждаю соответствующие сведения указаны ниже:

ПРИЛОЖЕНИЕ 2. АНКЕТА

нижеследующем:

Я, [ΦUO], настоящим подтверждаю, что мне известны Требования, содержащиеся в Уставе

Фонда, и на дату настоящий Анкеты заявляю о

4. After making reasonable inquiries on the below matters, I confirm that I am not a person with whom the Foundation or Yandex Group is restricted from having dealings due to restrictions imposed by laws applicable to the Foundation or any person within Yandex Group:

confirm / do not confirm, see	details
below:	

4.	В результате принятия разумных усилий для
	выяснения всех соответствующих
	обстоятельств, указанных ниже, я
	подтверждаю, что не являюсь лицом, с
	которыми Фонду или Группе Яндекс может
	быть запрещено сотрудничать в силу
	ограничений, наложенных правом,
	применимым к Фонду или любому лицу
	Группы Яндекс:
	П полтверждаю / П не полтверждаю.

подтверждаю	, L II	с подтверждаю
соответствующие	свеления	указаны ниже:
		<i>J</i> ====================================

- 5. After taking reasonable inquiries on the below matters, I confirm that I do not have currently and have not had within the preceding two (2) full calendar years commercial conflict of interest, whereas:
 - 5.1 commercial conflict of interest is defined as any commercial relationship between any business that operates in any area/line of activity (save for the areas/lines of activities in academic science, education and not-for-profit medicine), which represents above one per cent. (1%) of the gross revenue of the Yandex Group based on the consolidated financial statements prepared under generally accepted accounting principles in the US (US GAAP) for the six-month period ending on the last balance sheet date, as determined in accordance with US GAAP, and:
 - (i) myself:
 - (ii) my close relatives (including my spouse, parents, spouse's parents, children, siblings) or any person sharing my household;

- 5. В результате принятия разумных усилий для выяснения всех соответствующих обстоятельств, указанных ниже, я подтверждаю, что у меня отсутствует на текущий момент и отсутствовал в течение последних 2 (двух) полных календарных лет коммерческий конфликт интересов, при этом:
 - 5.1 коммерческий конфликт интересов определяется как наличие коммерческих отношений между бизнесом, который работает в любом секторе/сфере деятельности (за исключением секторов/сфер академической науки, образования и некоммерческой медицины), которая составляет более 1% (одного процента) в общей выручке Группы Яндекс по консолидированной отчетности Группы Яндекс, подготовленной по общепринятым стандартам бухгалтерской отчетности США (US GAAP), за шестимесячный период, оканчивающийся в последнюю дату баланса, как она определяется в соответствии с US GAAP, и:
 - (i) мной;
 - (ii) моими близкими родственниками (включая, моих супругу/супруга, родителей, родителей супруга/супруги, детей, братьев или сестер) или иными лицами, проживающими совместно со мной;

	(iii)	by companies directly or indirectly controlled by me or my close relatives;		(iii)	компаниями прямо или косвенно контролируемыми мной или моими близкими родственниками;
	(iv)	entities in which I or my close relatives have a shareholding of more than one per cent. (1%) of a listed company or three per cent. (3%) of a non-public company; and		(iv)	компаниями, в которых я или мои близкие родственники владеют пакетом акций более 1% (одного процента) в публичной компании или 3% (трех процентов) в непубличной компании; и
5.2	follor relati comp	mercial relationships include the wing direct or indirect onships with a business that betes with the business of lex Group:	5.2	себя косво котој	мерческие отношения включают в следующие прямые или енные отношения с бизнесом, рый конкурирует с бизнесом пы Яндекс:
	(i)	employment, including in the past;		(i)	трудовые отношения, в том числе в прошлом;
	(ii)	directorship, including in the past;		(ii)	членство в совете директоров, в том числе в прошлом;
	(iii)	consultancy arrangement;		(iii)	оказание услуг в качестве консультанта;
	(iv)	shareholding (more than one per cent. (1%) of a listed company or three per cent. (3%) of a non-public company)		(iv)	владение акциями (более 1% (одного процента) в публичной компании или 3% (трех процентов) в непубличной компании)
	confirn ow:	n / do not confirm, see details			ерждаю / П не подтверждаю, твующие сведения указаны ниже:
are true. The above s	he copi tateme	oregoing statements made by me es of the documents confirming nts are attached. I agree that I ny of the above statements are	сделанные документов заявления,	мной, в, подт прила ность	что вышеизложенные заявления, являются достоверными. Копии гверждающие вышеизложенные гаются. Я соглашаюсь нести за недостоверность указанных
Signature /	Подпи	ісь:			
[Full name					
Date / Дата	a:				
_		_			

SCHEDULE 3. CONSENT

ПРИЛОЖЕНИЕ 3. СОГЛАСИЕ

To	: [Foundation]	Ко	Кому: [Фонду]		
1.	I, [full name], hereby consent to hold the position of the Director [B/C] of the Foundation if appointed.	1.	Я, [ΦHO], настоящим подтверждаю свое согласие на занятие должности Директора [B/C] Φ онда в случае моего назначения.		
2.	This Consent is issued for an indefinite period and I may withdraw it by notifying the Foundation at the following email address:	2.	Настоящее Согласие выдано на неопределенный срок и может быть отозвано мной путем уведомления Фонда по адресу:		
Sig	gnature / Подпись:				
[Fi	ull name] / [ΦИО]				
Da	ta / Hara:				

SCHEDULE 4. COMPETENCE OF THE BOARD AND DECISION-MAKING PROCEDURE

- 1. The Board's scope of competence shall include adopting resolutions on the below matters, subject to the procedures set forth below.
- 2. The following special rules of decisionmaking procedure apply to the matters that in accordance with this Schedule 4 shall be decided by a qualified majority of votes of Directors A:
 - (i) each Director A has five (5) votes and each members of the Board of another class has one vote; and
 - (ii) the decision on such matter is adopted, provided that at least two Directors A cast all of their votes in favor of such decision.

 $[English\ version\ follows]$

ПРИЛОЖЕНИЕ 4. КОМПЕТЕНЦИЯ СОВЕТА ФОНДА И ПОРЯДОК ПРИНЯТИЯ РЕШЕНИЙ

- 1. К компетенции Совета Фонда относится принятие решений по нижеуказанным вопросам с соблюдением указанного порядка принятия решений.
- 2. К вопросам, для принятия решения по которым в соответствии с настоящим Приложением 4 требуется квалифицированное большинство голосов Директоров А, применяются следующий специальный порядок принятия решений:
 - (i) каждый Директор А обладает 5 (пятью) голосами, и каждый член Совета Фонда другого класса обладает одним голосом; и
 - (ii) решение считается принятым, если за него всеми своими голосами проголосовали как минимум два Директора А.

No.	Вопрос, по которому осуществляется голосование	Порядок принятия решения
Вопр	осы деятельности Фонда	
1.	Внесение изменений и дополнений в настоящий Устав	Единогласно всеми 11 членами Совета Фонда
2.	Решение о ликвидации Фонда	Единогласно всеми 11 членами Совета Фонда
3.	Решение о выплате членам Совета Фонда вознаграждения за исполнение возложенных на них обязанностей	Квалифицированное большинство голосов Директоров А
4.	(а) Утверждение финансового плана (бюджета) Фонда на финансовый год, включая утверждение размера компенсации расходов членам Совета Фонда в соответствии с п. 6.14 Устава, внесение изменений в финансовый план (бюджет);	Квалифицированное большинство голосов Директоров А
	(б) Одобрение сделок, перечисленных в п. 8.5 Устава	Как минимум большинство голосов Директоров А и большинство голосов Директоров В

No.	Вопрос, по которому осуществляется голосование	Порядок принятия решения
5.	Избрание Исполнительного директора из числа Директоров А и досрочное прекращение его полномочий	Квалифицированное большинство голосов Директоров А
6.	Утверждение годового отчета и бухгалтерской (финансовой) отчетности Фонда	Квалифицированное большинство голосов Директоров А
7.	Решение о прекращении полномочий члена Совета Фонда в случаях, указанных в п. 6.6.7 Устава	Единогласно всеми действующими членами Совета Фонда (голос члена Совета Фонда, в отношении которого проводится голосование, не учитывается)
8.	Решение о соответствии Требованиям в отношении кандидата на должность Директора С в порядке, предусмотренном п. 6.10.1 Устава	Как минимум два голоса Директоров А и как минимум два голоса Директоров В
9.	Решение об предоставлении согласия на назначение кандидата на должность Директора С в порядке, предусмотренном п. 6.10.2 Устава	Как минимум два голоса Директоров А плюс как минимум два голоса Директоров В
Вопр	осы, связанные с осуществлением Фондом прав уча	стия в иных лицах
10.	В порядке реализации прав по Специальной акции:	
	(а) решение о соответствии кандидатов, предлагаемых членами Совета Фонда к выдвижению на должность Специального директора, а также замещающего его лица требованиям Устава Учредителя	Как минимум пять голосов Директоров В плюс один голос Директора А или Директора С
	(б) выдвижение в порядке Обязывающей номинации кандидата на должность Специального директора, а также замещающего его лица в соответствии с процедурой, установленной Уставом Учредителя (в том числе, в случае, если лицо, ранее назначенное на должность Специального директора, стало неспособно осуществлять полномочия Специального директора по каким-либо причинам или прекратило свои полномочия) и решение о предоставлении полномочий Директору В на уведомление Учредителя о такой Обязывающей номинации, а также о выдаче Специальному директору инструкции о подаче заявления о прекращении своих полномочий	Как минимум пять голосов Директоров В плюс один голос

No.		прос, по которому осуществляется осование	Порядок принятия решения
	директоров Учредителя на принятие Советом		Как минимум пять голосов Директоров В плюс один голос Директора А или Директора С
 (г) предоставление предварительного согласия на использование Специальным директором, входящим в состав Комитета по назначениям, права вето по вопросу одобрения какой-либо кандидатуры, предлагаемой к избранию на должность директора Класса А в совет директоров Учредителя в соответствии с внутренними документами Учредителя (д) решение о предоставлении согласия потенциальному приобретателю акций Учредителя свыше Лимита владения согласно Статье 4С Устава Учредителя (е) решение о направлении требования Учредителю о созыве внеочередного общего собрания акционеров Учредителя, внесении в повестку дня общего собрания акционеров Учредителя вопроса о прекращении полномочий Специального директора и предоставлении полномочий Директору В на осуществление права голоса по указанному вопросу на общем собрании акционеров 		использование Специальным директором, входящим в состав Комитета по назначениям, права вето по вопросу одобрения какой-либо кандидатуры, предлагаемой к избранию на должность директора Класса А в совет директоров Учредителя в соответствии с	Как минимум пять голосов Директоров В плюс один голос Директора А или Директора С
		потенциальному приобретателю акций Учредителя свыше Лимита владения согласно	Как минимум пять голосов Директоров В плюс один голос Директора А или Директора С
		Учредителю о созыве внеочередного общего собрания акционеров Учредителя, внесении в повестку дня общего собрания акционеров Учредителя вопроса о прекращении полномочий Специального директора и предоставлении полномочий Директору В на осуществление права голоса по указанному	Как минимум пять голосов Директоров В плюс один голос Директора А или Директора С
	(ж)	предоставление согласия по запросу Совета директоров Учредителя на принятие Советом директоров Учредителя решения об одобрении кандидата для включения с Список кандидатов в случае, если на такой момент времени в Совет директоров Учредителя не входит ни один Специальный директор	Как минимум пять голосов Директоров В плюс один голос Директора А или Директора С
11.	B n	орядке реализации прав по Специальной доле:	
	(a)	решение о направлении уведомления Учредителю и ООО Яндекс о наличии основания для возникновения Особой корпоративной ситуации (в соответствии с тем, как основание возникновения Особой	Как минимум пять голосов Директоров В плюс один голос Директора А или Директора С

No.		прос, по которому осуществляется осование	Порядок принятия решения
		корпоративной ситуации определено в Приложении 5 к Уставу)	
	(6)	решение о наступлении Особой корпоративной ситуации и решение о созыве общего собрания участников ООО Яндекс по вопросам (i) прекращения полномочий Генерального директора по Дополнительному основанию, являющемуся Особой корпоративной ситуацией, и (ii) избрания Временного Генерального директора в соответствии с Уставом ООО Яндекс	Как минимум пять голосов Директоров В плюс один голос Директора А или Директора С
	(B)	определение варианта голосования Специальной долей на общем собрании участников ООО Яндекс по вопросу о прекращении полномочий Генерального директора по Дополнительным основаниям, указанным в пункте (б) выше, и о назначении Временного Генерального директора в связи с Особой корпоративной ситуацией	Как минимум пять голосов Директоров В плюс один голос Директора А или Директора С
	(r)	решение о направлении уведомления о возникновении Особой ситуации Учредителю и ООО Яндекс и определении срока для устранения Особой ситуации, который не может быть менее, чем 7 (семь) календарных дней	Как минимум пять голосов Директоров В плюс два голоса, каждый из которых должен быть голосом Директора А или Директора С
истече устран собран (i) пре директ являю назнач директ		решение о наличии Особой ситуации по истечении установленного срока для ее устранения и решение о созыве общего собрания участников ООО Яндекс по вопросам (i) прекращения полномочий Генерального директора по Дополнительному основанию, являющему Особой ситуацией, и (ii) назначения Временного Генерального директора в соответствии с Уставом ООО Яндекс	Как минимум пять голосов Директоров В плюс два голоса, каждый из которых должен быть голосом Директора А или Директора С
	(e)	определение варианта голосования Специальной долей на общем собрании участников ООО Яндекс по вопросу о прекращении полномочий Генерального Директора по Дополнительным основаниям, указанным в пункте (д) выше, и о назначении Временного Генерального директора в связи с Особой ситуацией	Как минимум пять голосов Директоров В плюс два голоса, каждый из которых должен быть голосом Директора А или Директора С

No.	Вопрос, по которому осуществляется голосование			Порядок принятия решения
	(ж)	Специ Генер	еление варианта голосования нальной долей по вопросу избрания нального директора (пункт 12.2.16 Устава Яндекс)	Простое большинство действующих членов Совета Фонда
	для включения в Список кандидатов и направления такого предложения Учредителю и ООО Яндекс и о варианте голосования на общем собрании участников ООО Яндекс по		ключения в Список кандидатов и вления такого предложения Учредителю О Яндекс и о варианте голосования на	Как минимум пять голосов Директоров В плюс один голос
	(и)	вопро	рение соглашения с Учредителем по су о прекращении Особой корпоративной ции или Особой ситуации	Как минимум пять голосов Директоров В плюс один голос Директора А или Директора С
	(к)	Яндек	ние о подаче заявления о выходе из ООО с или направления требования о бретении ООО Яндекс Специальной доли	Единогласно всеми 11 членами Совета Фонда
			и доли) в уставном капитале ООО Яндекс,	Как минимум пять голосов Директоров В плюс один голос
	(м) определение варианта голосования Специальной долей по иным вопросам общего собрания участников ООО Яндекс, принятие решений по которым требует единогласия в соответствии с Уставом ООО Яндекс, в том числе:		иальной долей по иным вопросам общего ния участников ООО Яндекс, принятие ний по которым требует единогласия в етствии с Уставом ООО Яндекс, в том	
		(i)	возложение дополнительных обязанностей на всех участников ООО Яндекс	Как минимум два голоса Директоров А плюс как минимум три голоса Директоров В
	(ii) прекращение дополнительных обязанностей участника ООО Яндекс (iii) предоставление дополнительных прав участнику (участникам) ООО Яндекс (iv) прекращение или ограничение дополнительных прав, предоставленных всем участникам ООО Яндекс		* *	Как минимум два голоса Директоров А плюс как минимум три голоса Директоров В
				Как минимум два голоса Директоров А плюс как минимум три голоса Директоров В
			дополнительных прав, предоставленных всем участникам	Как минимум два голоса Директоров А плюс как минимум три голоса Директоров В
		(v)	решение о возложении обязанности по внесению вкладов в имущество ООО Яндекс в порядке, отличающемся от	Как минимум два голоса Директоров А плюс как минимум три голоса Директоров В

No.	Вопрос, по голосован	о которому осуществляется ие	Порядок принятия решения
		порядка, установленного в пункте 10.3 Устава ООО Яндекс	
	(vi)	утверждение устава ООО Яндекс, внесение в него изменений или утверждение устава в новой редакции, принятие решения о том, что ООО Яндекс в дальнейшем действует на основании типового устава, либо о том, что ООО Яндекс в дальнейшем не будет действовать на основании типового устава, изменение размера уставного капитала ООО Яндекс, наименования ООО Яндекс, места нахождения ООО Яндекс	Как минимум два голоса Директоров А плюс как минимум три голоса Директоров В
	(vii)	решение о реорганизации или ликвидации ООО Яндекс	Как минимум два голоса Директоров А плюс как минимум три голоса Директоров В
	(viii)	решение о распределении доли в уставном капитале ООО Яндекс, принадлежащей ООО Яндекс	Как минимум два голоса Директоров А плюс как минимум три голоса Директоров В
	(ix)	решение о выплате кредиторам действительной стоимости доли или части доли участника в уставном капитале ООО Яндекс, на имущество которого обращается взыскание	Как минимум два голоса Директоров А плюс как минимум три голоса Директоров В
	(x)	утверждение формы и условий договора с Временным Генеральным директором	Как минимум два голоса Директоров А плюс как минимум три голоса Директоров В
	(xi)	утверждение денежной оценки неденежных вкладов в уставный капитал ООО Яндекс	Как минимум два голоса Директоров А плюс как минимум три голоса Директоров В
	(xii)	решение о распределении прибыли участникам ООО Яндекс непропорционально их долям в уставном капитале ООО Яндекс	Как минимум два голоса Директоров А плюс как минимум три голоса Директоров В

No.	Вопрос, по которому осуществляется голосование	Порядок принятия решения			
Проч	Ірочие вопросы				
12.	Решение об одобрении сделки, в которой имеется заинтересованность				
	(а) в случае заинтересованности Директора В или Директора С	Большинство голосов незаинтересованных членов Совета Фонда, включая как минимум два голоса Директоров А			
	(б) в случае заинтересованности Директора A, за исключением сделок, указанных в п. 10.6	Большинство голосов незаинтересованных членов Совета Фонда			
13.	Решение о прекращении полномочий Директора А в случае, предусмотренном п. 6.12.1 Устава	Большинство голосов Директоров В			
14.	Одобрение изменений в корпоративный договор, заключенный Фондом и Учредителем в отношении ООО Яндекс, и предоставлении полномочий Директору В на подписание дополнительных соглашений к такому корпоративному договору	Большинство голосов Директоров В			
15.	Решение об обращении с исковым заявлением к Учредителю	Как минимум пять голосов Директоров В плюс как минимум два голоса Директоров С			
16.	Принятие решений об утверждении и изменении Списка нотариусов	Большинство голосов Директоров А и большинство голосов Директоров В			
17.	Принятие решений по иным вопросам в соответствии с настоящим Уставом и Действующим законодательством	Большинство голосов Директоров А и большинство голосов Директоров В			

No.	Matter Put to Vote Procedures to Adopt Resolution					
Foun	undation's operational matters:					
1.	Amendment and modification of this Charter	Unanimous vote of all 11 members of the Board				
2.	Resolution on liquidation of the Foundation	Unanimous vote of all 11 members of the Board				
3.	Decision on payment of remuneration to the Board members in connection with exercising of their duties	Qualified majority of votes of Directors A				
4.	(a) Approval of the Foundation's financial plan (budget) for the financial year, including approval of compensation payable to the members of the Board for their expenses under paragraph 6.14 of the Charter, amendments to the financial plan (budget)	Qualified majority of votes of Directors A				
	(b) Approval of transactions referred to in paragraph 8.5	At least majority of votes of Directors A and majority of votes of Director B				
5.	Election of the Executive Director from among Directors A and early termination of such Executive Director's powers Qualified majority of votes of Directors A					
6.	Approval of the annual report and accounts (financial statements) of the Foundation	Qualified majority of votes of Directors A				
7.	Resolution on termination of a member of the Board in the circumstances set forth in paragraph 6.6.7 of the Charter Unanimous vote by all incumbent members of the Board (provided to the vote of the Board member who voted on shall not be taken account.)					
8.	Resolution on whether a candidate proposed for the position of Director C satisfies the Requirements, in accordance with paragraph 6.10.1 of the Charter	At least two votes of Directors A and at least two votes of Directors B				
9.	Resolution on granting a consent to appointment of a candidate to the position of Director C in accordance with paragraph 6.10.2 At least two votes of Directors A at least two votes of Directors B					
Matte	ers related to exercising the Foundation's right to partic	ipate in other entities				
10.	In connection with the Priority Share					
	(a) Resolution on whether candidates proposed by the Board members for the position of a Designated Director, as well as their substitutes, meet the requirements of the Founder's Articles	At least five votes of Directors B plus one vote of Director A or Director C				

No.	Ma	tter Put to Vote	Procedures to Adopt Resolution	
	(b)	Making a Binding Nomination of a Designated Director, as well as it substitute, in accordance with the procedures stipulated by the Founder's Articles (including in the event that a person appointed earlier to serve as a Designated Director becomes unable to serve as a Designated Director for any reason, or is terminated) and resolution on authorizing Director B to notify the Founder of such Binding nomination, as well as on instructing a Designated Director to submit a resignation	At least five votes of Directors B plus one vote	
	(c)	Granting an approval to the Board of Directors of the Founder, as may be requested thereby, to proceed on any matter not in compliance with a recommendation by the Public Policy Committee or in absence of a recommendation by the Public Policy Committee, where such recommendation is required under the Founder's internal regulation	At least five votes of Directors B plus one vote of Director A or Director C	
	(d)	Granting a prior approval to a Designated Director on the Nominating Committee to veto any nominee to the position of Class A director on the Founder's board of directors in accordance with the Founder's internal regulations	At least five votes of Directors B plus one vote of Director A or Director C	
	(e)	Granting an approval to a potential acquirer of shares in the Founder in excess of the Ownership Cap in accordance with Article 4C of the Founder's Articles	At least five votes of Directors B plus one vote of Director A or Director C	
	(f)	Resolution on submitting to the Founder of a request to convene an extraordinary general meeting of shareholders of the Founder, proposal to include into the agenda of the general meeting of the shareholders of the Founder the matter of terminating the powers of a Designated Director and authorizing Director B to exercise voting rights on such matter at the general meeting of the shareholders of the Founder	At least five votes of Directors B plus one vote of Director A or Director C	
	(g)	Granting an approval to the Board of Directors of the Founder, as may be requested thereby, to proceed on approval of the candidate(s) to be included into the List of Candidate, to the extent no Designated Directors are on the Board of Directors of the Founder	At least five votes of Directors B plus one vote of Director A or Director C	
11.	In c	connection with the Special Interest:		
	(a)	Resolution on sending a Special Corporate Situation notice to the Founder and Yandex LLC on the occurrence of a trigger for a Special	At least five votes of Directors B plus one vote of Director A or Director C	

No.	Ma	tter Put to Vote	Procedures to Adopt Resolution
		Corporate Situation (in accordance with the description of the trigger for a Special Corporate Situation specified in <u>Schedule 5</u> hereto)	
	(b)	Resolutions on occurrence of a Special Corporate Situation and calling a general meeting of participants of Yandex LLC in order (i) to terminate General Director due to an Additional Ground for termination that constitutes a Special Corporate Situation and (ii) to appoint Interim General Director in accordance with the Yandex LLC's Charter	At least five votes of Directors B plus one vote of Director A or Director C
	(c)	Determining how to vote the Special Interest at the general meetings of participants of Yandex LLC on termination of General Director on Additional Grounds referred to in item (b) above and on appointing Interim General Director in connection with a Special Corporate Situation	At least five votes of Directors B plus one vote of Director A or Director C
	(d)	Resolutions on sending a notice of Special Situation occurrence to the Founder and Yandex LLC and determining a time period within which such Special Situation must be rectified (provided that such time period may be not less than seven (7) calendar days)	At least five votes of Directors B plus two votes, each of which shall be a vote of Director A or Director C
	(e)	Resolutions on existence of a Special Situation upon the expiry of the time period by which such Special Situation was to be rectified and calling a General Meeting of Participants of Yandex LLC in order (i) to terminate General Director on Additional Ground for termination that constitutes a Special Situation and (ii) to appoint Interim General Director in accordance with the Yandex LLC's Charter	At least five votes of Directors B plus two votes, each of which shall be a vote of Director A or Director C
	(f)	Determining how to vote the Special Interest at the General Meetings of Participants of Yandex LLC on termination of General Director on Additional Grounds referred to in item (e) above and on appointing Interim General Director in connection with an Special Situation	At least five votes of Directors B plus two votes, each of which shall be a vote of Director A or Director C
	(g)	Determining how to vote the Special Interest on the matter of election of the General Director (clause 12.2.16 of the Yandex LLC's Charter)	Simple majority vote of incumbent members of the Foundation Board
	(h)	Resolution on the matter of proposing candidates for inclusion in the List of Candidates and submitting such proposal on amending the List of Candidates to the Founder and Yandex LLC	At least five votes of Directors B plus one vote

No.	Ma	tter Pu	t to Vote	Procedures to Adopt Resolution
	(i)	concer	val of the agreement with the Founder rning elimination of a Special Corporate ton or a Special Situation	At least five votes of Directors B plus one vote of Director A or Director C
	(j)			Unanimous vote of all 11 members of the Board
	(k)	partici	ng Foundation's consent for the pledge of patory interest (or part thereof) held by the er in the charter capital of Yandex LLC	At least five votes of Directors B plus one vote
	(1)	other i of Yar adopte	mining how to vote the Special Interest on matters of the general meeting of participants and LLC that require resolutions to be do by a unanimous vote in accordance with undex LLC's Charter, including:	
		(i)	imposition of additional obligations on all participants of Yandex LLC	At least two votes of Directors A plus at least three votes of Directors B
		(ii)	termination of additional obligations of a participant of Yandex LLC	At least two votes of Directors A plus at least three votes of Directors B
		(iii)	granting additional rights to a participant (participants) of Yandex LLC	At least two votes of Directors A plus at least three votes of Directors B
		(iv)	termination or limitation of additional rights granted to all participants of Yandex LLC	At least two votes of Directors A plus at least three votes of Directors B
		(v)	decision to oblige the participants of Yandex LLC to make contributions to the Yandex LLC's assets other than in accordance with the procedures sets forth in Section 10.3 of the Yandex LLC's Charter	At least two votes of Directors A plus at least three votes of Directors B
		(vi)	approving the Yandex LLC's Charter, amending it or approving a new version of the Yandex LLC's Charter, making a decision that Yandex LLC will continue to operate on the basis of the model charter, or that Yandex LLC will not continue to operate on the basis of the model charter, changing the amount of the charter capital of Yandex LLC, changing the name of Yandex LLC or its registered address;	At least two votes of Directors A plus at least three votes of Directors B
		(vii)	decision on the reorganization or liquidation of Yandex LLC	At least two votes of Directors A plus at least three votes of Directors B

No.	Matter Pu	t to Vote	Procedures to Adopt Resolution
	(viii)	decision on allocation of the share in the charter capital of Yandex LLC owned by Yandex LLC	At least two votes of Directors A plus at least three votes of Directors B
	(ix)	decision on payment to creditors of the actual value of a share in the charter capital of Yandex LLC owned by a participant of Yandex LLC whose assets are being foreclosed	At least two votes of Directors A plus at least three votes of Directors B
	(x)	approval of the form and terms and conditions of the agreement with Interim General Director	At least two votes of Directors A plus at least three votes of Directors B
	(xi)	approval of the monetary value of non- monetary contributions to the assets of Yandex LLC made by the participants of Yandex LLC	At least two votes of Directors A plus at least three votes of Directors B
	(xii)	decision on distribution of profit among the Participants disproportionately to the sizes of their shares in the charter capital of Yandex LLC	At least two votes of Directors A plus at least three votes of Directors B
Other	r Matters		
12.	Approval o	f an interested party transaction	
	(a) if Dire	ector B or Director C is interested	Majority of votes of the members of the Board who are not interested, including at least two votes of Directors A
		ector A is interested, other than in relation to ections referred to in paragraph 10.6	Majority of votes of the members of the Board who are not interested
13.		n terminating the powers of Director A in with paragraph 6.12.1 of the Charter	Majority of votes of Directors B
14.	entered into relation to	f amendments to the corporate agreement between the Foundation and the Founder in Yandex LLC and authorizing Director B to y such amendments	Majority of votes of Directors B
15.	Resolution	to raise a claim against the Founder	At least five votes of Directors B plus at least two votes of Directors C
16.	Approval o	f the List of Notaries	Majority of votes of Directors A and majority of votes of Director B
17.	Resolutions and Applica	s on other matters envisaged by this Charter able Law	Majority of votes of Directors A and majority of votes of Director B

SCHEDULE 5. SPECIFIC REQUIREMENTS TO DECISION-MAKING

ПРИЛОЖЕНИЕ 5. ОСОБЫЙ ПОРЯДОК ПРИНЯТИЯ РЕШЕНИЙ

The Board shall be entitled to make a decision on the occurrence of a Special Corporate Situation (per item 11(b) of <u>Schedule 4</u>) or a Special Situation (per item 11(e) of <u>Schedule 4</u>) provided that all of conditions specified in Part 1 or in Part 2 of this <u>Schedule 5</u>, as applicable, have been fully met.

Совет Фонда правомочен принимать решения по вопросу о наступлении Особой корпоративной ситуации (п. 11(б) Приложения 4 к Уставу) или о наступлении Особой ситуации (п. 11(д) Приложения 4 к Уставу) только при соблюдении условий, указанных в Части 1 или Части 2 настоящего Приложения 5, соответственно.

[English version follows]

№	Усло	вия
Час	ть 1. Р	ешение о наступлении Особой корпоративной ситуации
1.	(a)	Основание возникновения Особой корпоративной ситуации: Комитет по общественным интересам не сформирован; и
	(б)	Совет Фонда правомочен принять решение о наступлении Особой корпоративной ситуации, если выполнены все нижеуказанные условия:
		(i) на очередном заседании Совета директоров Учредителя, но в любом случае не позднее, чем по истечении 4 (четырех) месяцев после основания возникновения Особой корпоративной ситуации по п. 1(а), не принято решение о формировании Комитета по общественным интересам; и
		(ii) Совет Фонда принял решение, предусмотренное п. 11(a) <u>Приложения 4</u> к Уставу, в связи с основанием возникновения Особой корпоративной ситуации, указанным в п. 1(a), и Фонд уведомил Учредителя о таком принятом решении с указанием на конкретное основание возникновения Особой корпоративной ситуации (при условии, что на дату такого уведомления Комитет по общественным интересам все еще не сформирован); и
		(iii) Совет директоров Учредителя не принял решение о формировании Комитета по общественным интересам на следующем очередном заседании Совета директоров Учредителя, проведенном после получения Учредителем уведомления Фонда, указанного в п. 1(б)(ii) выше, но в любом случае не позднее, чем по истечении 4 (четырех) месяцев после получения Учредителем уведомления Фонда, указанного в п. 1(б)(ii) выше.
	(B)	Особая корпоративная ситуация по основанию, указанному в настоящем п. 1, считается прекращенной в дату, в которую: (i) Совет директоров Учредителя принял решение о формировании Комитета по общественным интересам; или (ii) Учредитель и Фонд пришли к соглашению, что такая Особая корпоративная ситуация прекратилась.

№	Услов	
2.	(a)	Основание возникновения Особой корпоративной ситуации: Совет циректоров Учредителя принял решение о роспуске Комитета по общественным интересам; и
	(б)	Совет Фонда правомочен принять решение о наступлении Особой сорпоративной ситуации, если выполнены все нижеуказанные условия:
		Совет Фонда принял решение, предусмотренное п. 11(а) <u>Приложения 4</u> к Уставу, в связи с основанием возникновения Особой корпоративной ситуации, указанным в п. 2(а), и Фонд уведомил Учредителя о таком принятом решении с указанием на конкретное основание возникновения Особой корпоративной ситуации (при условии, что на дату такого уведомления Комитет по общественным интересам все еще не сформирован); и
		і) решение о формировании Комитета по общественным интересам не принято на следующем очередном заседании Совета директоров Учредителя, проведенном после получения Учредителем уведомления Фонда, указанного в п. 2(б)(і) выше, но в любом случае не позднее, чем по истечении 4 (четырех) месяцев после получения Учредителем уведомления, указанного в п. 2(б)(і).
	(B)	Особая корпоративная ситуация по основанию, указанному в настоящем п. 2, читается прекращенной в дату, в которую: (i) Совет директоров Учредителя принял решение о формировании Комитета по общественным интересам; или ii) Учредитель и Фонд пришли к соглашению, что такая Особая сорпоративная ситуация прекратилась.
3.	(a)	Основание возникновения Особой корпоративной ситуации: Совет циректоров Учредителя не принял решение о включении любого из Специальных директоров в состав Комитета по назначениям; и
	(б)	Совет Фонда правомочен принять решение о наступлении Особой сорпоративной ситуации, если выполнены все нижеуказанные условия:
		Совет Фонда принял решение, предусмотренное п. 11(а) <u>Приложения 4</u> к Уставу, в связи с основанием возникновения Особой корпоративной ситуации, указанным в п. 3(а), и Фонд уведомил Учредителя о таком принятом решении с указанием на конкретное основание возникновения Особой корпоративной ситуации (при условии, что на дату такого уведомления ни один Специальный директор не был включен в состав Комитета по назначениям); и
		решение о включении Специального Директора в состав Комитета по назначениям не принято на следующем очередном заседании Совета директоров Учредителя, проведенном после получения Учредителем уведомления Фонда, указанного в п. 3(б)(i) выше, но в любом случае не позднее, чем по истечении 4 (четырех) месяцев после получения Учредителем уведомления, указанного в п. 3(б)(i) выше.
	(в)	Особая корпоративная ситуация по основанию, указанному в настоящем п. 3, очитается прекращенной в дату, в которую: (i) Совет директоров Учредителя принял решение о включении Специального директора в состав Комитета по назначениям; или (ii) Учредитель и Фонд пришли к соглашению, что такая Особая корпоративная ситуация прекратилась.

№	Условия		
4.	(a)	Основание возникновения Особой корпоративной ситуации: общее собрание акционеров Учредителя отклонило Обязывающую номинацию Фонда в отношении любого из Специальных директоров; и	
	(б)	Совет Фонда правомочен принять решение о наступлении Особой корпоративной ситуации, если выполнены все нижеуказанные условия:	
		(i) Совет Фонда принял решение, предусмотренное п. 11(a) <u>Приложения 4</u> к Уставу, в связи с основанием возникновения Особой корпоративной ситуации, указанным в п. 4(a), и Фонд уведомил Учредителя о таком принятом решении с указанием на конкретное основание возникновения Особой корпоративной ситуации (при условии, что на дату такого уведомления общим собранием акционеров Учредителя не была утверждена Обязывающая номинация в отношении любого из Специальных директоров); и	
		(ii) Фонд в соответствии с настоящим Уставом принял решение о предложении в порядке Обязывающей номинации кандидата на должность Специального директора и уведомил Учредителя о принятом решении в соответствии с процедурой, предусмотренной Уставом Учредителя и внутренними документами Учредителя; и	
		(iii) Обязывающая номинация Фонда в отношении Специального директора, указанная в п. 4(б)(ii), не утверждена общим собранием акционеров Учредителя в течение 60 (шестидесяти) дней после получения Учредителем уведомления Фонда, указанного в п. 4(б)(i) выше.	
	(B)	Особая корпоративная ситуация по основанию, указанному в настоящем п. 4, считается прекращенной в дату, в которую: (i) Обязывающая номинация Фонда в отношении кандидата на должность Специального директора утверждена общим собранием акционеров Учредителя; или (ii) Учредитель и Фонд пришли к соглашению, что такая Особая корпоративная ситуация прекратилась.	
5.	(a)	Основание возникновения Особой корпоративной ситуации: общее собрание акционеров Учредителя приняло решение о прекращении полномочий любого из Специальных директоров (за исключением случая, если такое решение принято с согласия Фонда); и	
	(б)	Совет Фонда правомочен принять решение о наступлении Особой корпоративной ситуации, если выполнены все нижеуказанные условия:	
		(i) Совет Фонда принял решение, предусмотренное п. 11(a) <u>Приложения 4</u> к Уставу, в связи с основанием возникновения Особой корпоративной ситуации, указанным в п. 5(a), и Фонд уведомил Учредителя о таком принятом решении с указанием на конкретное основание возникновения Особой корпоративной ситуации (при условии, что на дату такого уведомления общим собранием акционеров Учредителя не была утверждена Обязывающая номинация в отношении соответствующего Специального директора); и	
		(ii) Фонд в соответствии с настоящим Уставом принял решение о предложении в порядке Обязывающей номинации кандидата на должность Специального директора и уведомил Учредителя о	

№	Условия		
			принятом решении в соответствии с процедурой, предусмотренной Уставом и внутренними документами Учредителя; и
		(iii)	Обязывающая номинация Фонда в отношении кандидата на должность Специального директора, указанная в п. 5(б)(ii), не утверждена общим собранием акционеров Учредителя в течение 60 (шестидесяти) дней после получения Учредителем уведомления Фонда, указанного в п. 5(б)(ii) выше.
	(B)	счи: Фон утве Фон	бая корпоративная ситуация по основанию, указанному настоящем в п. 5, гается прекращенной в дату, в которую: (i) Обязывающая номинация нда в отношении кандидата на должность Специального директора ерждена общим собранием акционеров Учредителя; или (ii) Учредитель и нд пришли к соглашению, что такая Особая корпоративная ситуация кратилась.
6.	(a)	акці дир Учр назн осно дир	кование возникновения Особой корпоративной ситуации: общее собрание ионеров Учредителя приняло решение о назначении на позицию ектора класса I (как этот термин определен во внутренних документах редителя) кандидата, не рекомендованного подкомитетом I Комитета по начениям ("Несоответствующий кандидат") (во избежание сомнений, ованием возникновения не будет являться ситуация, когда такая позиция ектора класса I остаётся вакантной после проведения указанного выше цего собрания акционеров Учредителя); и
	(б)		ет Фонда правомочен принять решение о наступлении Особой поративной ситуации, если выполнены все нижеуказанные условия:
		Уст ситу при Осо	ет Фонда принял решение, предусмотренное п. 11(a) Приложения 4 к аву, в связи с основанием возникновения Особой корпоративной уации, указанным в п. 6(a), и Фонд уведомил Учредителя о таком нятом решении с указанием на конкретное основание возникновения бой корпоративной ситуации не позднее, чем за 90 (девяносто) дней доы проведения очередного общего собрания акционеров Учредителя; и
		пол	дующее очередное общее собрание акционеров Учредителя после учения Учредителя уведомления Фонда, указанного в п. 6(б)(i), не няло решение:
			 о назначении кандидата, соответствующего рекомендации подкомитета I Комитета по назначениям ("Соответствующий кандидат") на позицию директора, являющегося директором класса II; или о прекращении полномочий Несоответствующего кандидата в качестве директора класса I, и в результате этого позиция директора класса I осталась вакантной либо на место Несоответствующего кандидата в качестве директора класса I назначен Соответствующий
		П	кандидат.
		Учр	целей п. 6(б)(ii) следующим очередным общим собранием акционеров едителя после получения Учредителем уведомления, указанного в п. (i) считается:
			в случае если очередное общее собрание акционеров Учредителя приходится на период после истечения 90 (девяноста) дней с даты получения Учредителем уведомления Фонда, указанного в п. 6(б)(i) выше,

№	Услов	Условия			
		 – ближайшее годовое общее собрание акционеров Учредителя; и 			
		 в случае если очередное общее собрание акционеров Учредителя приходится на период до истечения 90 (девяноста) дней с даты получения Учредителем уведомления Фонда, указанного в п. 6(б)(i) выше, – годовое общее собрание акционеров Учредителя в следующем календарном году. 			
	(B)	Особая корпоративная ситуация по основанию, указанному в настоящем п. 6, считается прекращенной в дату, которую:			
	(i)	общее собрание акционеров Учредителя приняло решение о назначении Соответствующего кандидата на позицию директора, являющегося директором класса II; или			
	(ii)	общее собрание акционеров Учредителя приняло решение о прекращении полномочий Несоответствующего кандидата в качестве директора класса I, и в результате этого позиция директора класса I осталась вакантной либо на место Несоответствующего кандидата в качестве директора класса I назначен Соответствующий кандидат; или			
	(iii)	Учредитель и Фонд пришли к соглашению, что такая Особая корпоративная ситуация прекратилась.			
7.	(a)	Основание возникновения Особой корпоративной ситуации: ООО Яндекс действует без согласия общего собрания участников по вопросам, предусмотренным пп. 12.2.31-12.2.34 Устава ООО Яндекс, или Учредитель голосует по таким вопросам на общем собрании участников ООО Яндекс не в соответствии с ранее принятым решением Комитета по общественным интересам или в отсутствие решения Комитета по общественным интересам об одобрении соответствующего действия (сделки) ООО Яндекс, при условии, что Фонд не предоставил Учредителю своё согласие в порядке пункта 10(в) Приложения 4 настоящего Устава; и			
	(6)	Совет Фонда правомочен принять решение о наступлении Особой корпоративной ситуации, если выполнены все нижеуказанные условия:			
		(i) Совет Фонда принял решение, предусмотренное п. 11(a) <u>Приложения 4</u> к Уставу, в связи с основанием возникновения Особой корпоративной ситуации, указанным в п. 7(a), и Фонд уведомил Учредителя и ООО Яндекс и о таком принятом решении с указанием конкретных обстоятельств, связанных с несоблюдением ограничений, наложенных пп. 12.2.31-12.2.34 Устава ООО Яндекс, и действия, необходимого для устранения основания Особой корпоративной ситуации, которое должно заключаться в совершении любого из следующих действий:			
		(х) получении одобрения Комитета по общественным интересам или Фонда в порядке пункта 10(в) Приложения 4 настоящего Устава на такое действие (сделку) в соответствии с учредительными и внутренними документами Учредителя; или			
		(у) в отношении ограничений, предусмотренных пп. 12.2.31-12.2.32 и 12.2.34 Устава ООО Яндекс:			
		 отмене ООО Яндекс действия (сделки); или 			
		 оспаривании ООО Яндекс соответствующего действия (сделки) и подаче искового заявления в компетентный суд 			

№	Услов	Условия		
			(третейский суд) о признании сделки недействительной и (или) применении последствий недействительности сделки при условии, что такое исковое заявление было принято к производству компетентным судом (третейским судом)); или	
			(z) в отношении ограничений, предусмотренных п. 12.2.33 Устава ООО Яндекс: принятии соответствующей внутренней политики ООО Яндекс и его Контролируемых лиц в предыдущей редакции, действовавшей до изменения, ставшего основанием возникновения Особой корпоративной ситуации;	
			(при условии, что на дату такого уведомления такое действие не выполнено); и	
		(ii)	в течение 60 (шестидесяти) календарных дней с момента получения Учредителем и ООО Яндекс уведомления Фонда, указанного в п. 7(б)(i) выше, не выполнено любое действие, необходимое для устранения основания Особой корпоративной ситуации в соответствии с п. 7(б)(i) выше.	
	(r)		н корпоративная ситуация по основанию, указанному в настоящем п. 7, тся прекращенной:	
	•		, в которую выполнено действие, необходимое для устранения ания Особой корпоративной ситуации в соответствии с п. 7(б)(i) выше;	
	•	полно Особо выше,	ае, если Временный Генеральный директор, действуя в рамках своих мочий, не выполнил действия, необходимые для устранения основания й корпоративной ситуации в соответствии с пп. $7(\delta)(i)(x) - 7(\delta)(i)(z)$ в течение 60 (шестидесяти) календарных дней с даты его избрания — в стечения такого 60-дневного срока; или	
	•		, в которую Учредитель и Фонд пришли к соглашению, что такая я корпоративная ситуация прекратилась.	
Час	ть 2. Ре	шение о	наступлении Особой ситуации	
8.			правомочен принять решение о наступлении Особой ситуации, если е нижеуказанные условия:	
	(a)	предус решени ситуац состав. Учреди	Фонда принял решение о возникновении Особой ситуации в порядке, мотренном п. 11(г) Приложения 4 к Уставу, в том числе, в таком ии указано, какое нарушение привело к возникновению Особой ии, и определен срок для устранения Особой ситуации, который ляет не менее 7 (семи) календарных дней, и уведомил ООО Яндекс и ителя о таком решении с приложением нотариально удостоверенной такого решения; и	
	(б)		н ситуация связана с вопросами национальной безопасности Российской ации; и	
	(B)	решен	ечении срока для устранения Особой ситуации, установленного ием Совета Фонда, указанным в п. 8(а) выше, соответствующая Особая ия не устранена.	
			едомил ООО Яндекс и Учредителя о наличии Особой ситуации, как о п. 8(a), то Фонд не вправе повторно принимать решение о	

№	Условия	•
	возникновении Особой ситуации в отношении того же самого нарушения.	

No.	Descri	ption		
Part	Part 1. Decision on occurrence of a Special Corporate Situation			
1.	(a)	Trigger for the Special Corporate Situation: the Public Policy Committee is not formed; and		
	(b)	The Board is entitled to make a decision on occurrence of the Special Corporate Situation, if all of the below conditions are met:		
		(i) on regular meeting, but in any event not later than four (4) months after the trigger for the Special Corporate Situation referred to in section 1(a) has occurred, the Board of Directors of the Founder has failed to adopt a decision to form the Public Policy Committee; and		
the Charter, in connection referred to in section 1(a) such decision and specific (provided that as of the date of		the Board has made a decision set forth by section 11(a) of <u>Schedule 4</u> to the Charter, in connection to the trigger for a Special Corporate Situation referred to in section 1(a), and the Foundation has notified the Founder of such decision and specified the trigger for the Special Corporate Situation (provided that as of the date of such notification the Public Policy Committee has not been formed yet); and		
		(iii) the Board of Directors of the Founder has failed to adopt a decision to form the Public Policy Committee at the next regular meeting of the Board of Directors of the Founder, held after the receipt by the Founder of the notification from the Foundation, described in section 1(b)(i) above, but in any event not later than four (4) months after the receipt by the Founder of the notification from the Foundation, described in section 1(b)(i) above.		
	(c)	A Special Corporate Situation with the trigger in this section 1 shall be deemed eliminated on the date, when: (i) the Board of Directors of the Founder adopted a decision to form the Public Policy Committee; or (ii) the Founder and the Foundation agreed that this Special Corporate Situation is eliminated.		
2.	(a)	Trigger for the Special Corporate Situation: the Board of Directors of the Founder has made a decision to dismiss the Public Policy Committee; and		
	(b)	The Board is entitled to make a decision on occurrence of the Special Corporate Situation, if all of the below conditions are met:		
		(i) the Board has made a decision forth by section 11(a) of <u>Schedule 4</u> to the Charter, in connection to the trigger for a Special Corporate Situation, referred to in section 3(a) and the Foundation has notified the Founder of such taken decision and specified the trigger for the Special Corporate Situation (provided that as of the date of such notification the Public Policy Committee still has not been formed yet); and		
		(ii) the decision to form the Public Policy Committee is not made at the next ordinary meeting of the Board of Directors of the Founder, which is held after the receipt by the Founder of the notification from the Foundation, described in section 2(b)(i) above, but in any event not later than four (4) months after the receipt by the Founder of the notification from the Foundation, described in section 2(b)(i) above.		
	(c)	A Special Corporate Situation with the trigger in this section 2 shall be deemed eliminated on the date, when: (i) the Board of Directors of the Founder has adopted a decision to form the Public Policy Committee; or (ii) the Founder and Foundation agreed that this Special Corporate Situation is eliminated.		

No.	Descri	Description			
3.	(a)	Trigger for the Special Corporate Situation: the Board of Directors of the Founder has failed to adopt a decision to include any one Designated Director in the Nominating Committee; and			
	(b)	The Board is entitled to make a decision on occurrence of the Special Corporate Situation, if all of the below conditions are met:			
		(i) the Board has made a decision forth by section 11(a) of <u>Schedule 4</u> to the Charter, in connection to the trigger for a Special Corporate Situation, listed in section 3(a) and the Foundation has notified the Founder of such taken decision and specified the trigger for the Special Corporate Situation (provided that as of the date of such notification no Designated Director has been appointed to the Nominating Committee); and			
		(ii) the decision to appoint a Designated Director to the Nominating Committee is not made at the next ordinary meeting of the Board of Directors of the Founder, which is held after the receipt by the Founder of the notification described in section 3(b)(i) above from the Foundation, but in any event not later than four (4) months after the receipt by the Founder of the notification described in section 3(b)(i) above.			
	(c)	A Special Corporate Situation with the trigger in this section 3 shall be deemed eliminated on the date, when: (i) the Board of Directors of the Founder has adopt a decision to include a Designated Director in the Nominating Committee; or (ii) Founder and Foundation agreed that this Special Corporate Situation is eliminated.			
4.	(a)	Trigger for the Special Corporate Situation: the general shareholders meeting of the Founder has rejected the Binding Nomination of the Foundation with respect to any of the Designated Directors; and			
	(b)	The Board is entitled to make a decision on occurrence of the Special Corporate Situation, if all of the below conditions are met:			
in d		the Board has made a decision forth by section 11(a) of <u>Schedule 4</u> to the Charter, in connection to the trigger for a Special Corporate Situation, listed in section 4(a) and the Foundation has notified the Founder of such taken decision and specified the trigger for the Special Corporate Situation (provided that as of the date of such notification the general shareholders meeting of the Founder has not approved the Binding Nomination with respect to any of the Designated Directors); and			
		(ii) the Foundation has, in accordance with this Charter, made a decision on offering in by way of a Binding Nomination of candidate to the position of the Designated Director and has notified the Founder of the taken decision in accordance with the procedure, set forth in the Founder's Articles and the internal documents of the Founder; and			
		the Binding Nomination of the Foundation with respect to the Designated Director described in section 4(b)(ii) has not been approved by the general shareholders meetings of the Founder within sixty (60) days after receipt by the Founder of the notification by the Foundation described in section 4(b)(i).			
	(c) A Special Corporate Situation with the trigger in this section 4 shall be deen eliminated on the date, when: (i) the Binding Nomination of the Foundation respect to the candidate to the position of the Designated Directors has been approved by the general shareholders meetings of the Founder; or (ii) Foundation				

No.	Description				
		Foundation agreed that this Special Corporate Situation is eliminated.			
5.	(a)	Trigger for the Special Corporate Situation: the general shareholders meeting of the Founder has taken a decision to terminate the authorities of any of the Designated Directors (other than in case such decision has been made upon the Foundation's consent); and			
	(b)	The Board is entitled to make a decision on occurrence of the Special Corporate Situation, if all of the below conditions are met:			
		(i) the Board has made a decision forth by section 11(a) of <u>Schedule 4</u> to the Charter, in connection to the trigger for a Special Corporate Situation, listed in section 5(a) and the Foundation has notified the Founder of such taken decision (provided that as of the date of such notification the general shareholders meeting of the Founder has not approved the Binding Nomination with respect to the relevant Designated Director); and			
		(ii) the Foundation has, in accordance with this Charter, adopted a decision to make a Binding Nomination in respect of a candidate to the position of Designated Directors and has notified the Founder of the taken decision in accordance with the procedure set forth by the Founder's Articles and internal documents of the Founder and specified the trigger for the Special Corporate Situation; and			
		(iii) the Binding Nomination of the Foundation in respect of the candidate to the position of the Designated Director described in section 5(b)(ii) has not been approved by the general shareholders meeting of the Founder within sixty (60) days after receipt by the of the notification by the Foundation, described in section 5(b)(i) above.			
eliminated on the date, when: (i) the F respect to the candidate to the position approved by the general shareholders		A Special Corporate Situation with the trigger in this section 5 shall be deemed eliminated on the date, when: (i) the Binding Nomination of the Foundation with respect to the candidate to the position of the Designated Director has been approved by the general shareholders meeting of the Founder; or (ii) Founder and Foundation agreed that this Special Corporate Situation is eliminated.			
6.	(a)	Trigger for the Special Corporate Situation the general shareholders meeting of the Founder has decided to fill the vacancy of class I director (as defined in the internal documents of the Founder) with a candidate who was not recommended by subcommittee I of the Nominating Committee ("Noncompliant Candidate") (for avoidance of doubt, the situation where the position of class I director remains vacant as a result of such general shareholders meeting of the Founder is not considered a trigger); and			
	(b)	The Board is entitled to make a decision on occurrence of the Special Corporate Situation, if all of the below conditions are met:			
		(i) the Board has made a decision set forth by section 11(a) of <u>Schedule 4</u> to the Charter, in connection with the trigger for a Special Corporate Situation, listed in section 6(a) and the Foundation has notified the Founder of such taken decision and specified the trigger for the Special Corporate Situation not later than ninety (90) days prior to the date of the regular general shareholders meeting of the Founder; and			
	(ii) the next regular general shareholders meeting of the Founder after the receipt by the Founder of the notification of the Foundation referred to in				

No.	Descri	ption		
	section 6(b)(i) failed to:			
		-	appoint to a candidate recommended by subcommittee I of the the Nominating Committee (" Compliant Candidate ") to the position of class II director, and, as a result, the relevant position of class I director remains vacant or a Compliant Candidate is appointed to replace the Noncompliant Candidate holding the position class I director; or	
	 terminate the authority of the Noncompliant Candidate as class I of 			
	For the purposes of section 6(b)(ii), the next regular general shareholders me of the Founder after the receipt by the Founder of the notification of the Foundation described in section 6(b)(i) above is:			
		-	to the extent the regular general shareholders meeting of the Founder falls on any date upon expiry of the ninety (90) days' period following the receipt by the Founder of the notification of the Foundation described in section 6(b)(i) above – the earliest regular general shareholders meeting of the Founder; and	
		-	to the extent the regular general shareholders meeting of the Founder falls on any date within falls on a date within the ninety (90) days' period following the receipt by the Founder of the notification of the Foundation described in section 6(b)(i) above, – the regular general shareholders meeting taking place in the following calendar year.	
	(c)	A Special Corporate Situation with the trigger in this section 6 shall be deel eliminated on the date, when:		
		(i)	the general shareholders meeting of the Founder made a decision to appoint Compliant Candidate to the position of class II director; or	
		(ii)	the general shareholders meeting of the Founder made a decision to authority of the Noncompliant Candidate as class I director, and and, as a result, the relevant position of class I director remains vacant or a Compliant Candidate is appointed to replace the Noncompliant Candidate holding the position class I director; or	
		(iii)	Founder and Foundation agreed that this Special Corporate Situation is eliminated.	
7.	(a)	of a General Meeting on issues set forth in sections 12.2.31 – 12.2.34 of the Yar LLC's Charter, or the Founder votes on those issues not in accordance with the earlier adopted decision of the Public Policy Committee, or in the absence of a decision of the Public Policy Committee or the Fund granting its consent as stipulated by section 10(c) of Appendix 4 on approval of a respective action (transaction) of Yandex LLC; and		
	(b)			
		(i)	the Board has made a decision forth by section 11(a) of Appendix 4 to the Charter, in connection to the trigger for a Special Corporate Situation, listed in section 7(a) and the Foundation has notified the Founder and Yandex LLC of such taken decision and specified the circumstances that resulted in the failure to comply with the restrictions imposed by sections 12.2.31 – 12.2.34 of the Yandex LLC's Charter and the actions that need to be taken	

No.	Description			
	to eliminate the trigger for the Special Corporate Situation, which must be one of the following actions:			
	(x) obtaining decision of the Public Policy Committee or the Fund as stipulated by section 10(c) of Appendix 4 on approval of such action (transaction) in accordance with the constituent and internal documents of Founder, or			
	(y) with respect to restrictions imposed by sections 12.2.31-12.2.32 and 12.2.34 of the Yandex LLC Charter:			
	 cancellation by Yandex LLC of the action (transaction) or 			
	 challenging by Yandex LLC of the relevant action (transaction) and filing a claim to a competent court (arbitration tribunal) to invalidate the transaction or to apply the consequences of invalidity, provided such claim has been accepted by a competent court (arbitration tribunal); 			
	(z) with respect to restrictions imposed by section 12.2.33 of Yandex LLC Charter: approval (reinstatement) of the relevant internal policy of Yandex LLC or its Controlled persons as in effect before its amendment, if such amendment served as a trigger for the Special Corporate Situation;			
	(provided that as of the date of such notification none of the relevant action specified above has not been made); and			
	(ii) within sixty (60) days after receipt by the Founder and Yandex LLC of the notification by the Foundation, described in section 7(b)(i) above, none of the relevant actions, which is necessary for eliminating the Special Corporate Situation and which is specified in section 7(b)(i) above has been made.			
	(c) A Special Corporate Situation with the trigger in this section 7 shall be deemed eliminated:			
	• on the date, when the relevant action necessary for elimination of the Special Corporate Situation has been made as specified in section 7(b)(i) above, or			
	• if the Interim General Director, acting within its authority, has failed to make actions necessary to eliminate the Special Corporate Situation specified in items 7(b)(i)(x)-7(b)(i)(z) during sixty (60) calendar days from the appointment of such Interim General Director – on the date of the expiry of such 60-day term; or			
	 on the date, when Founder and Foundation agreed that this Special Corporate Situation is eliminated. 			
Part	2. Decision on the occurrence of a Special Situation			
8.	The Board is entitled to make a decision on occurrence of the Special Situation, if all of the below conditions are met:			
	(a) the Board has made a decision forth by section 11(e) of <u>Schedule 4</u> to the Charter, including has specified in such decision which violation resulted in the occurrence of the Special Situation and a time period for such Special Situation to be rectified, which in no case may be less than seven (7) calendar days, and the Foundation has notified the Founder and Yandex LLC of such decision with a notarized copy of such decision enclosed; and			

No. Description (b) the Special Situation relates to matters of the national security of the Russian Federation; and (c) upon expiry of the period to cure the Special Situation, set forth by the decision of the Board described in section 8(a) above, the respective Special Situation is not cured. In the event the Foundation has notified the Founder and Yandex LLC about the existence of the Special Situation as set out in section 8(a), the Foundation shall not be entitled to adopt another decision on the occurrence of the Special Situation in connection with the same violation.

SCHEDULE 6. FORM OF THE POWER OF ATTORNEY

[Удостоверительная надпись нотариуса]

ПРИЛОЖЕНИЕ 6. ФОРМА ДОВЕРЕННОСТИ

ДОВЕРЕННОСТЬ

[Дата] [Место]

Я, [ФИО], паспорт гражданина РФ № [\bullet], выдан [когда] г. [кем], код подразделения [\bullet], зарегистрированный по адресу: [\bullet] ("Доверитель"), настоящей доверенностью назначаю и уполномочиваю:

[ФИО]

(паспорт гражданина РФ № [\bullet], выдан [∂ama] г. [κeM], код подразделения [\bullet], зарегистрированного по адресу: г. [\bullet]);

- 1. представлять Доверителя в качестве члена Совета Фонда Директора [указать класс] [Международного Фонда Общественных интересов], основной государственный регистрационный номер (ОГРН) [●] ("Фонд"), и от имени Доверителя по своему усмотрению осуществлять право голоса Доверителя на заседаниях Совета Фонда, проводимых в течение срока действия настоящей Доверенности.
- 2. подписывать, получать и вручать все документы, выполнять все процедуры и формальности, а также совершать все действия, которые могут оказаться необходимыми или целесообразными в связи с голосованием на заседании Совета Фонда.
- 3. Представлять Доверителя в качестве члена Совета Фонда в отношениях с Фондом, другими членами Совета Фонда, нотариусами по вопросам, связанным с осуществлением права голоса Доверителя на заседаниях Совета Фонда в течение срока действия Доверенности.

Настоящая доверенность выдана сроком до [указать срок не более 30 (тридцати) календарных дней].

Подпись:		
[ΦИΟ]		